

RIO ALTO WATER DISTRICT COMMUNITY FACILITIES DISTRICT NO. 2011-1 (WASTEWATER SYSTEM IMPROVEMENT PROJECT)

CFD TAX ADMINISTRATION REPORT FISCAL YEAR 2023-24

September 27, 2024

Community Facilities District No. 2011-1 CFD Tax Administration Report

TABLE OF CONTENTS

Sect	tion	<u>Page</u>
	Executive Summary	i
I.	Introduction	1
II.	Purpose of Report	3
III.	Fiscal Year 2023-24 Special Tax Requirement	4
IV.	Fiscal Year 2023-24 Special Tax Levy	5
V.	Parcel Classification and EDU Assignment for FY 2023-24	7
VI.	Prepayments	9
VII.	State Reporting Requirements	10
Appe	endix A – Summary of Fiscal Year 2023-24 Special Tax Levy	
Appe	endix B – Fiscal Year 2023-24 Special Tax Levy for Individual Assessor's Parcel	S
Appe	endix C - Rate and Method of Apportionment of Special Tax	
Anne	endix D – Boundary Man of CFD No. 2011-1	

The following summary provides a brief overview of the main points from this report regarding the Rio Alto Water District Community Facilities District No. 2011-1 ("CFD No. 2011-1" or the "CFD"):

Fiscal Year 2023-24 Special Tax Levy

Number of Taxed Parcels	Total Special Tax Levy
1,356	\$341,280

For further detail regarding the special tax levy, or special tax rates, please refer to Section IV of this report.

Parcel and Equivalent Dwelling Unit ("EDU") Summary for Fiscal Year 2023-24

Property Classification	Total Number of Parcels	Total Number of EDUs
Single EDU Parcel	1,292	1,292
Duplex	16	32
Triplex	1	3
Fourplex	0	0
Other Multiple EDU Parcel	42	89
Taxable POA Property	5	6
Optional LPSS/Septic Property	38	0
Tract 1017, Phase 2 Property	248	0
Total	1,642	1,422

For more information regarding the classification of parcels and assignment of EDUs in CFD No. 2011-1, please see Section V of this report.

Delinquency Summary

Delinquent Amount for FY 2023-24 (as of May 7, 2024)	Number of Parcels Delinquent	Total Levy for FY 2023-24	Delinquency Rate
\$29,520	137	\$341,280	8.65%

Rio Alto Water District Community Facilities District No. 2011-1

On January 11, 2012, the Board of Directors ("Board") of the Rio Alto Water District ("Water District") adopted Resolution No. 02-12 to establish CFD No. 2011-1 and to call for an election to be held on April 17, 2012. In that election, the registered voters within the CFD voted to authorize the levy of a Mello-Roos special tax on property within CFD No. 2011-1. The registered voters also approved the CFD to incur bonded indebtedness in an amount not to exceed \$6,215,000. The Board adopted Resolution No. 08-12 to declare the election results on April 30, 2012. Furthermore, the Board passed Ordinance No. 100-1 on May 23, 2012, to levy a special tax within the CFD beginning in fiscal year 2012-13. The Board approved Resolution 12-12 on August 31, 2012, authorizing staff to execute documents to obtain a loan from the United States Department of Agriculture-Rural Development ("USDA-RD") in an amount not to exceed \$5,000,000. Finally, on December 19, 2012, the Board adopted Resolution No. 14-12, dedicating CFD special tax revenues to payment of any Clean Water State Revolving Fund ("CWSRF") loans. The CWSRF loan in the total principal amount of \$1,215,000 closed on May 13, 2013. Subsequently, the \$5,000,000 USDA-RD loan closed on August 15, 2014. Special tax revenues from taxable property in the CFD are used to secure the USDA-RD and CWSRF loans.

The proceeds from the USDA-RD and CWSRF loans will be used to pay for certain authorized facilities and expenses. These authorized facilities include various improvements and additions to the Water District's wastewater system in order to provide for the treatment of sewage and transmission and discharge of treated wastewater onto approximately 78 acres of constructed wetlands. These improvements address requirements of the California Toxics Rule as detailed in the California Regional Water Quality Control Board's Waste Discharge Requirements Order No. R5-2010-0103 and Cease and Desist Order No. R5-2010-0104. These improvements will result in the Water District ceasing regular discharge to the Sacramento River and discharging to land only via the constructed wetlands, except during extreme peak flows in excess of the 100-year event.

The CFD is located adjacent to the Sacramento River in the County of Tehama ("County"). It is approximately 23 miles south of the city of Redding and 150 miles north of the city of Sacramento. CFD No. 2011-1 includes approximately 1,700 parcels of both single and multifamily residential property in varying degrees of development.

The Mello-Roos Community Facilities Act of 1982

The reduction in property tax revenue that resulted from the passage of Proposition 13 in 1978 required public agencies and real estate developers to look for other means to fund public infrastructure. The funding available from traditional assessment districts was limited by certain requirements of the assessment acts, and it became clear that a more flexible funding tool was needed. In response, the California State Legislature ("Legislature") approved the Mello-Roos Community Facilities Act of 1982 ("Act"), which provides for the levy of a special tax within a defined geographic area, namely a community facilities district, if such a levy is approved by

two-thirds of the qualified electors in the area. Community facilities districts can generate funding for a broad range of facilities, and special taxes can be allocated to property in any reasonable manner other than on an ad valorem basis.

Authorized Facilities

The special tax revenues from CFD No. 2011-1 are authorized to fund certain wastewater infrastructure improvements. Wastewater improvements include the following:

- i. Installation of a selector and screening unit;
- ii. Construction and installation of a secondary clarifier and improvements to upgrade the existing secondary clarifier and effluent pumps;
- iii. Construction and installation of a return activated sludge (RAS) pump station and conversion of the existing RAS pump station into an aerobic digester;
- iv. Electrical improvements at the wastewater treatment plant;
- v. Construction and installation of approximately 11,600 linear feet of 10-inch chlorination force main;
- vi. Acquisition of land and development thereon of constructed wetlands; and
- vii. Appurtenant work and improvements to the existing wastewater system of the Water District in order to provide for the collection and treatment of sewage and the transmission and discharge of treated wastewater effluent onto the constructed wetlands.

These improvements address requirements of the California Toxics Rule and will result in the Water District ceasing regular discharge to the Sacramento River and discharging to land only via the constructed wetlands, except during extreme peak flows in excess of the 100-year event.

II. PURPOSE OF REPORT

This CFD Tax Administration Report ("Report") presents findings from research and financial analysis performed by Goodwin Consulting Group, Inc. to determine the fiscal year 2023-24 special tax levy for CFD No. 2011-1. The Report is intended to provide information to interested parties regarding the current financial obligations of the CFD, special taxes to be levied in fiscal year 2023-24, and public facilities authorized to be funded by the CFD. The Report also summarizes the classification of parcels well as other pertinent information (e.g., prepayments and delinquencies) for CFD No. 2011-1.

The Report is organized into the following sections:

- Section III identifies financial obligations of the CFD for fiscal year 2023-24.
- <u>Section IV</u> provides a summary of the methodology that is used to apportion the special tax among parcels in the CFD.
- Section V summarizes the classification of parcels and assignment of EDUs in CFD No. 2011-1.
- Section VI identifies parcels, if any, that have prepaid their special tax obligation.
- Section VII provides information on state reporting requirements.

III. FISCAL YEAR 2023-24 SPECIAL TAX REQUIREMENT

Pursuant to the Rate and Method of Apportionment of Special Tax ("RMA") for the CFD, the Special Tax Requirement means, for any fiscal year, the total of: (i) principal and interest due, including scheduled sinking fund payments, on bonds, loans or other debt, (ii) administrative expenses for such fiscal year, (iii) an amount needed to create or replenish reserve funds, (iv) an amount needed to cure any delinquencies in the payment of principal and interest on indebtedness of the CFD which have occurred in the previous fiscal year or are anticipated to occur for the current fiscal year, and (v) pay-as-you-go expenditures for authorized facilities to be constructed or acquired by the CFD. The amounts referred to in the preceding sentence may be reduced in any fiscal year by collected penalties associated with delinquent special taxes, available earnings on or surpluses in funds for bonds, loans, or other debt, and any other revenues available to pay debt service. Furthermore, the amounts related to items (iii) and (iv) of the Special Tax Requirement are subject to the 10% limitation described in Section D of the RMA. For fiscal year 2023-24, the total Special Tax Requirement is \$341,280.

Assignment of EDUs

Special taxes within CFD No. 2011-1 are levied pursuant to the methodology set forth in the RMA. The RMA establishes various categories which define the circumstances necessary to assign an equivalent dwelling unit ("EDU") to each parcel. The special tax levied on a particular property is based on the number of EDUs that property is assigned. (All capitalized terms are defined in the RMA in Appendix C of this Report.)

Single EDU Parcels are parcels of taxable property that are assigned one EDU. Any properties with the following characteristics shall all be assigned one EDU:

- i. Residential Property on which one single family detached unit has been constructed.
- ii. Each parcel with a Halfplex.
- iii. Any parcel of taxable property that is a Buildable Lot on which no structure has yet been constructed.
- iv. Specific parcels of Taxable Property Owner Association Property, as identified in the RMA.

Multiple EDU Parcels are parcels of taxable property which are assigned two or more EDUs. Any properties with the following characteristics shall be assigned the corresponding number of EDUs:

- i. A parcel of Residential Property with a duplex shall be assigned two EDUs.
- ii. A parcel of Residential Property with a triplex shall be assigned three EDUs.
- iii. A parcel of Residential Property with a fourplex shall be assigned four EDUs.
- iv. Any parcel with a residential structure with five or more Residential Units shall be assigned a number of EDUs equal to the number of Residential Units on the parcel.
- v. Specific properties and merged lots are assigned two EDUs, as identified in the RMA.
- vi. If two or more parcels are combined to form one or more new parcels, the aggregate number of EDUs assigned to the original parcels shall be assigned to the newly-created parcel(s).
- vii. Any parcel of Residential Property with more than one Residential Unit may be a Multiple EDU Parcel, as determined by the Water District based on services requested for each Residential Unit.

viii. Parcels with commercial or other non-residential buildings shall be assigned two EDUs unless the Water District determines that a different number of EDUs is more appropriate based on the type of construction.

Maximum Special Tax Rates

The maximum annual special tax for each parcel of taxable property in the CFD is equal to \$289 per EDU. In addition, a one-time Catch-Up Special Tax may be levied on parcels of Optional LPSS Property or Optional Septic Property prior to approval by the Water District of the property owner's application for sewer service, as described in Section C.2 of the RMA.

Apportionment of Annual Special Taxes in Fiscal Year 2023-24

The actual annual special tax levied on each parcel in CFD No. 2011-1 each fiscal year is determined by application of Section D of the RMA. Pursuant to this section, the special tax shall be levied as follows until the amount of the levy is equal to the Special Tax Requirement for that fiscal year. The Special Tax Requirement is distributed among all parcels of taxable property in proportion to the number of EDUs assigned to each parcel, up to the maximum annual special tax for each such parcel.

For fiscal year 2023-24, a total of 1,422 EDUs were assigned to the various parcels within the CFD. Taxing these EDUs at 83.0% of their maximum annual special tax rates, \$240 per EDU, will generate approximately \$341,280 in special tax revenue, assuming no delinquencies. This amount is equal to the Special Tax Requirement. Appendix A shows a summary of the special tax levy for fiscal year 2023-24, and Appendix B includes the special tax levied on individual parcels in CFD No. 2011-1.

Based on the current parcels in CFD No. 2011-1, the following table summarizes the classification of parcels and assignment of EDUs within the CFD for the fiscal year 2023-24 special tax levy.

Community Facilities District No. 2011-1 Parcel and EDU Summary for Fiscal Year 2023-24

Property Classification	Total Number of Parcels	Total Number of EDUs
Single EDU Parcel	1,292	1,292
Duplex	16	32
Triplex	1	3
Fourplex	0	0
Other Multiple EDU Parcel	42	89
Taxable POA Property	5	6
Optional LPSS/Septic Property	38	0
Tract 1017, Phase 2 Property	248	0
Total	1,642	1,422

Several types of property shown in the above table are exempt, and therefore are not subject to the special tax levy. These types include Public Property as well as some properties that are given the choice to connect to the sewer system, but have not yet exercised that option. Exempt properties are further defined as follows:

- i. Public Property, except for Public Property that had, in its entirety, been a parcel of taxable property in a prior fiscal year.
- ii. Property Owner Association Property, except for certain parcels of Taxable POA Property and any buildable lot owned by the POA that is expected to be resold to an individual property owner.
- iii. Parcels of Optional LPSS Property. Once a parcel of Optional LPSS Property has connected to the Water District's sewer system, it shall thereafter be classified as taxable property.
- iv. Parcels of Optional Septic Property. Once a parcel of Optional Septic Property has connected to the Water District's sewer system, it shall thereafter be classified as taxable property.

- v. Parcels of Tract 1017, Phase 2 Property. All parcels of Tract 1017, Phase 2 Property are exempt until such time as the Water District accepts water and sewer improvements for property within Phase 2 of Tract 1017.
- vi. Parcels owned by a public utility for an unmanned facility.
- vii. Parcels subject to an easement that precludes any other use on the parcels.
- viii. Any parcel that has fully prepaid its annual special tax obligation.

VI. PREPAYMENTS

As of June 30, 2023, no property owner has prepaid his/her special tax obligation. Therefore, all parcels of taxable property remain subject to the annual special tax levy.

Senate Bill No. 165

On September 18, 2000, former Governor Gray Davis signed Senate Bill 165 which enacted the Local Agency Special Tax and Bond Accountability Act. In approving the bill, the Legislature declared that local agencies need to demonstrate to the voters that special taxes and bond proceeds are being spent on the facilities and services for which they were intended. To further this objective, the Legislature added Sections 50075.3 and 53411 to the California Government Code setting forth annual reporting requirements relative to special taxes collected and bonds issued by a local public agency. Pursuant to Sections 50075.3 and 53411, the "chief fiscal officer" of the public agency will, by January 1, 2002, and at least once a year thereafter, file a report with the City setting forth (i) the amount of special taxes that have been collected and expended; (ii) the status of any project required or authorized to be funded by the special taxes; (iii) if bonds have been issued, the amount of bonds that have been collected and expended; and (iv) if bonds have been issued, the status of any project required or authorized to be funded from bond proceeds.

Assembly Bill No. 1666

On July 25, 2016, former Governor Jerry Brown signed Assembly Bill No. 1666, adding Section 53343.2 to the California Government Code ("GC"). The bill enhances the transparency of community facilities districts by requiring that certain reports be accessible on a local agency's web site. Pursuant to Section 53343.2, a local agency that has a web site shall, within seven months after the last day of each fiscal year of the district, display prominently on its web site the following information:

Item (a): A copy of an annual report, if requested, pursuant to GC Section 53343.1. The report required by Section 53343.1 includes CFD budgetary information for the prior fiscal year and is only prepared by a community facilities district at the request of a person who resides in or owns property in the community facilities district. If the annual report has not been requested to be prepared, then a posting to the web site would not be necessary.

Item (b): A copy of the report provided to the California Debt and Investment Advisory Commission ("CDIAC") pursuant to GC Section 53359.5. Under Section 53359.5, local agencies must provide CDIAC with the following: (i) notice of proposed sale of bonds; (ii) annual reports on the fiscal status of bonded districts; and (iii) notice of any failure to pay debt service on bonds, or of any draw on a reserve fund to pay debt service on bonds.

Item (c): A copy of the report provided to the State Controller's Office pursuant to GC Section 12463.2. This section refers to the parcel tax portion of a local agency's Financial Transactions Report that is prepared for the State Controller's Office annually. Note that school districts are not subject to the reporting required by GC Section 12463.2.

Assembly Bill No. 1483

On October 9, 2019, Governor Gavin Newsom signed Assembly Bill No. 1483, adding Section 65940.1 to the California Government Code. The law requires that a city, county, or special district that has an internet website, maintain on its website a current schedule of fees, exactions, and affordability requirements imposed by the public agency on all housing development projects. Pursuant to Section 65940.1, the definition of an exaction includes a special tax levied pursuant to the Mello-Roos Community Facilities Act.

Assembly Bill No. 1483 defines a housing development project as consisting of (a) residential units only; or (b) mixed-use developments consisting of residential and non-residential land uses with at least two-thirds of the square footage designated for residential use; or (c) transitional housing or supportive housing. Assembly Bill No. 1483 also requires a city, county, or special district to update this information on their website within 30 days of any changes made to the information

APPENDIX A

Summary of Fiscal Year 2023-24 Special Tax Levy

Rio Alto Water District Community Facilities District No. 2011-1 (Wastewater System Improvement Project)

Special Tax Levy Summary for Fiscal Year 2023-24

Special Tax Category	Number of Parcels	Number of EDUs	FY 2023-24 Maximum Special Tax Rate	FY 2023-24 Actual Special Tax Rate	FY 2023-24 Special Tax Levy
Single EDU Parcel	1,292 Parcels	1,292 EDUs	\$289.00	\$240.00	\$310,080.00
Duplex	16 Parcels	32 EDUs	\$289.00	\$240.00	\$7,680.00
Triplex	1 Parcel	3 EDUs	\$289.00	\$240.00	\$720.00
Fourplex	0 Parcels	0 EDUs	\$289.00	\$0.00	\$0.00
Other Multiple EDU Parcel (2 Merged Lots)	36 Parcels	71 EDUs	\$289.00	\$240.00	\$17,040.00
Other Multiple EDU Parcel (3 Merged Lots)	5 Parcels	11 EDUs	\$289.00	\$240.00	\$2,640.00
Other Multiple EDU Parcel (7 Merged Lots)	1 Parcel	7 EDUs	\$289.00	\$240.00	\$1,680.00
Taxable POA Property	5 Parcels	6 EDUs	\$289.00	\$240.00	\$1,440.00
Optional LPSS/Septic Property	38 Parcels	0 EDUs	\$289.00	\$0.00	\$0.00
Tract 1017, Phase 2 Property	248 Parcels	0 EDUs	\$289.00	\$0.00	\$0.00
Total Fiscal Year 2023-24 Special Tax Levy					\$341,280.00

Goodwin Consulting Group, Inc.

APPENDIX B

Fiscal Year 2023-24 Special Tax Levy for Individual Assessor's Parcels

Assessor's Parcel Number	Property Classification	Number of EDUs	FY 2023-24 Maximum Special Tax	FY 2023-24 Special Tax Levy
000 000 000	ारता प्रदेश राज्य विकास स्थापन स्थापना स्थापना स्थापना स्थापना स्थापना स्थापना स्थापना स्थापना स्थापना स्थापना स्थापना स्थापना स्थापन		\$289.00	\$240.00
009-080-036-000	Taxable Property Owner Association Property	1	\$289.00 \$289.00	\$240.00
100-100-002-000	Single Equivalent Dwelling Unit		ي چار د در از د د د در و د د د در و د د د د د د د د د	\$240.00
100-100-003-000	Single Equivalent Dwelling Unit	Same and the standard addition of the	\$289.00 \$289.00	\$240.00
100-100-004-000	Single Equivalent Dwelling Unit	i makan dalah d T	ray practical representations of the second section of the second section of the second second second second section of the second seco	\$240.00
100-100-005-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00 \$240.00
100-100-006-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00 \$240.00
100-100-007-000	Single Equivalent Dwelling Unit	1	\$289.00	Market and the state of the sta
100-100-008-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-100-009-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-100-010-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00 \$240.00
100-100-011-000	Single Equivalent Dwelling Unit	The state of the s	\$289,00	\$240.00
100-100-012-000	Single Equivalent Dwelling Unit		\$289.00	Consider Checky Leving the Section of the Constitution of the Cons
100-100-013-000	Single Equivalent Dwelling Unit		\$289,00	\$240,00
100-100-014-000	Single Equivalent Dwelling Unit	A Tallegain Arthréigh (Acc) T	\$289.00	\$240.00
100-100-017-000	Single Equivalent Dwelling Unit		\$289.00 \$289.00	\$240.00
100-110-001-000	Single Equivalent Dwelling Unit	unitate (%sift) (25 filedii) T	and the second s	\$240.00
100-110-002-000	Single Equivalent Dwelling Unit	Over Sprinson	\$289,00	\$240.00
100-110-003-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-110-004-000	Single Equivalent Dwelling Unit	a and the section	\$289.00	\$240.00
100-110-005-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-110-006-000	Single Equivalent Dwelling Unit	aha ist <mark>i</mark> sibata	\$289,00	\$240.00
100-110-007-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-110-008-000	Single Equivalent Dwelling Unit		\$289,00	\$240,00
100-110-009-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-110-010-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-110-013-000	Single Equivalent Dwelling Unit	I MEMBANG PHINGS	\$289.00	\$240.00
100-110-014-000	Single Equivalent Dwelling Unit		\$289,00	\$240.00
100-110-015-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-110-016-000	Single Equivalent Dwelling Unit		\$289.00	\$240,00
100-110-017-000	Single Equivalent Dwelling Unit	1 	\$289.00	\$240.00
100-110-018-000	Single Equivalent Dwelling Unit	alian kabuta 1	\$289.00	\$240.00
100-110-019-000	Single Equivalent Dwelling Unit	l v ja popuja neganjerang oppaja saj	\$289,00	\$240.00
100-110-020-000	Single Equivalent Dwelling Unit		\$289.00	\$240,00
100-110-021-000	Single Equivalent Dwelling Unit	in - 14 kir milit ili kultur karan in kir I	\$289.00	\$240.00
100-110-022-000	Single Equivalent Dwelling Unit		\$289,00	\$240.00
100-110-023-000	Single Equivalent Dwelling Unit	1 	\$289,00	\$240.00
100-110-024-000	Single Equivalent Dwelling Unit		\$289,00	\$240.00
100-110-025-000	Single Equivalent Dwelling Unit	l metas rossen exemplement	\$289.00	\$240.00
100-110-026-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-110-027-000	Single Equivalent Dwelling Unit	1 kga makabananbasa	\$289.00	\$240.00
100-110-028-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-110-029-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240,00
100-120-001-000	Single Equivalent Dwelling Unit	1	\$289,00	\$240.00
100-120-002-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-120-003-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00

\$240.00 \$240.00 \$240.00 \$240.00 \$240.00 \$240.00 \$240.00
\$240.00 \$240.00 \$240.00 \$240.00 \$240.00
\$240.00 \$240.00 \$240.00 \$240.00
\$240.00 \$240.00 \$240.00
\$240.00 \$240.00
\$240.00
Constitution of the state of the second section in Section 19
4
\$240.00
\$240.00
\$240.00
\$240.00
\$240.00
\$240.00
\$240.00
\$240.00
\$240.00
\$240.00
\$240.00
\$240.00
\$240.00
\$240.00
\$240.00
\$240.00
\$240.00
\$240.00
\$240,00
\$240.00
\$240,00
\$240.00
\$240.00
\$240.00
\$240.00
\$240.00
\$240.00
\$240.00
\$240.00
\$0.00
\$240.00
\$240.00
\$240.00
\$240.00
\$240.00
\$240.00
J= 10.00
\$240.00
《香香·香香·香香·香香·香香·香香·香香·香香·香香·香香·香香·香香·香香·

Assessor's Parcel Number	Property Classification	Number of EDUs	FY 2023-24 Maximum Special Tax	FY 2023-24 Special Tax Levy
100-140-006-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-140-007-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-140-008-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-140-009-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-140-010-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-140-011-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-140-012-000	Single Equivalent Dwelling Unit	1	\$289,00	\$240.00
100-140-013-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-140-014-000	Single Equivalent Dwelling Unit		\$289.00	\$240,00
100-140-018-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-140-019-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-140-020-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-140-021-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-140-022-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-140-023-000	Single Equivalent Dwelling Unit	1	\$289,00	\$240.00
100-140-025-000	Other Multiple Equivalent Dwelling Units	1.5	\$433.50	\$360.00
100-140-026-000	Other Multiple Equivalent Dwelling Units	1.5	\$433.50	\$360.00
100-150-001-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-150-002-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-150-003-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240,00
100-150-004-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-150-005-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-150-006-000	Single Equivalent Dwelling Unit		\$289.00	\$240,00
100-150-007-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-150-008-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-150-009-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-150-010-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-150-011-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-150-012-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-150-013-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-150-014-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-150-015-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-150-016-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-150-017-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-150-020-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-150-021-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-150-022-000	Single Equivalent Dwelling Unit	1	\$289,00	\$240.00
100-150-023-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-150-024-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-150-025-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-150-026-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-150-027-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-150-028-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-150-029-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-150-031-000	Other Multiple Equivalent Dwelling Units	2	\$578.00	\$480.00

Assessor's Parcel Number	Property Classification	Number of EDUs	FY 2023-24 Maximum Special Tax	FY 2023-24 Special Tax Levy
100-160-001-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-160-002-000	Single Equivalent Dwelling Unit	i v	\$289,00	\$240.00
100-160-003-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-160-004-000	Single Equivalent Dwelling Unit	1	\$289,00	\$240,00
100-160-005-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240,00
100-160-006-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-160-007-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-160-008-000	Single Equivalent Dwelling Unit		\$289,00	\$240.00
100-160-011-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-160-012-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-160-013-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-160-014-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-160-015-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-160-016-000	Single Equivalent Dwelling Unit	1	\$289,00	\$240.00
100-160-017-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-160-018-000	Single Equivalent Dwelling Unit	1	\$289,00	\$240.00
100-160-019-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-160-020-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-160-021-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-160-022-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-160-023-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-160-024-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.90
100-160-025-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-160-026-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-170-003-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-170-004-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-170-005-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-170-006-000	Duplex	2	\$578.00	\$480.00
100-170-007-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-170-008-000	Duplex	2	\$578.00	\$480.00
100-170-009-000	Duplex	2	\$578.00	\$480.00
100-170-010-000	Single Equivalent Dwelling Unit	1 -	\$289.00	\$240,00
100-170-011-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-170-012-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-170-013-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-170-014-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-170-015-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-170-016-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-170-017-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-170-018-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-170-019-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-170-020-000	Single Equivalent Dwelling Unit		\$289,00	\$240.00
100-170-021-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-170-022-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-170-023-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00

Assessor's Parcel Number	Property Classification	Number of EDUs	FY 2023-24 Maximum Special Tax	FY 2023-24 Special Tax Levy
100-170-024-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-170-025-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-170-026-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-170-027-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-170-028-000	Single Equivalent Dwelling Unit	1	\$289,00	\$240.00
100-170-029-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-170-030-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-170-031-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-170-032-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-170-033-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-180-001-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-180-002-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-180-003-000	Single Equivalent Dwelling Unit		\$289,00	\$240,00
100-180-004-000	Single Equivalent Dwelling Unit	l	\$289.00	\$240.00
100-180-005-000	Single Equivalent Dwelling Unit		\$289,00	\$240.00
100-180-006-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-180-007-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-180-008-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-180-009-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-180-010-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240,00
100-180-011-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-180-012-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-180-013-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240,00
100-180-014-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-180-015-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-180-016-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-180-021-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-180-022-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-180-023-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-180-024-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-180-025-000	Single Equivalent Dwelling Unit	1 32	\$289.00	\$240.00
100-180-027-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-180-028-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-190-001-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-190-002-000	Single Equivalent Dwelling Unit	\mathbf{i}	\$289.00	\$240.00
100-190-003-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-190-004-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-190-007-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-190-008-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-190-009-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-190-010-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-190-011-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-190-012-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-190-013-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-190-014-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00

Assessor's Parcel Number	Property Classification	Number of EDUs	FY 2023-24 Maximum Special Tax	FY 2023-24 Special Tax Levy
100-190-015-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-190-016-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-190-017-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-190-018-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-190-019-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-190-020-000	Single Equivalent Dwelling Unit		\$289.00	\$240,00
100-190-023-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-190-024-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-190-025-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-190-026-000	Single Equivalent Dwelling Unit		\$289.00	\$240,00
100-190-027-000	Single Equivalent Dwelling Unit	• 1	\$289.00	\$240.00
100-190-028-000	Single Equivalent Dwelling Unit	1	\$289,00	\$240.00
100-190-029-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-190-031-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-190-032-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-190-033-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-190-034-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-190-035-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-190-036-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-190-037-000	Single Equivalent Dwelling Unit	1.5	\$289,00	\$240.00
100-190-039-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-190-040-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-190-041-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-240-009-000	Single Equivalent Dwelling Unit		\$289,00	\$240.00
100-240-010-000	Optional LPSS Property/Optional Septic Propert	v 0	\$0.00	\$0.00
100-240-019-000	Single Equivalent Dwelling Unit	i i i i i i i i i i i i i i i i i i i	\$289.00	\$240.00
100-240-020-000	Single Equivalent Dwelling Unit	i i i i i i i i i i i i i i i i i i i	\$289.00	\$240.00
100-240-021-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-240-034-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-240-035-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240,00
100-240-036-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-240-037-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-240-038-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-240-039-000	Optional LPSS Property/Optional Septic Proper	v 0	\$0.00	\$0.00
100-240-040-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-240-041-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240,00
100-240-042-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-250-001-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-250-001-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-250-002-000	Single Equivalent Dwelling Unit	.1	\$289.00	\$240.00
100-250-003-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-250-005-000	Single Equivalent Dwelling Unit	\mathbf{i}	\$289.00	\$240.00
100-250-006-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-250-007-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-250-008-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00

Assessor's Parcel Number	Property Classification	Number of EDUs	FY 2023-24 Maximum Special Tax	FY 2023-24 Special Tax Levy
100-250-009-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-250-010-000	Single Equivalent Dwelling Unit	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	\$289.00	\$240.00
100-250-011-000	Single Equivalent Dwelling Unit		\$289.00	\$240,00
100-250-012-000	Single Equivalent Dwelling Unit	ar 15° an 19° day da sakali kawalah daka sa 1	\$289.00	\$240.00
100-250-013-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-250-014-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-250-017-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-250-018-000	Single Equivalent Dwelling Unit	.i. Janu <u>1</u>	\$289.00	\$240.00
100-250-019-000	Single Equivalent Dwelling Unit		\$289.00	\$240,00
100-250-020-000	Single Equivalent Dwelling Unit	izatzista angaziraki kalenda Salenda-esek 1	\$289.00	\$240.00
100-250-021-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-250-022-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-250-023-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240,00
100-250-024-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-250-025-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-250-035-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-250-036-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-250-037-000	Single Equivalent Dwelling Unit	1	\$289,00	\$240.00
100-250-038-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-250-039-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-250-040-000	Single Equivalent Dwelling Unit	1	\$289,00	\$240.00
100-250-043-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-260-001-000	Single Equivalent Dwelling Unit	1	\$289,00	\$240,00
100-260-002-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240,00
100-260-003-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240,00
100-260-004-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-260-005-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-260-006-000	Single Equivalent Dwelling Unit	1	\$289,00	\$240.00
100-260-007-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-260-008-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-260-009-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-260-010-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-260-013-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-260-014-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-260-015-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240,00
100-260-016-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240,00
100-260-017-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-260-018-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-260-019-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-260-020-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-260-021-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-260-022-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-260-023-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240,00
100-260-026-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-260-027-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00

Assessor's Parcel Number	Property Classification	Number of EDUs	FY 2023-24 Maximum Special Tax	FY 2023-24 Special Tax Levy
100-260-028-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-260-029-000	Single Equivalent Dwelling Unit	$\mathbf{I}_{i}^{\mathbf{r}}$	\$289.00	\$240.00
100-260-030-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-260-034-000	Single Equivalent Dwelling Unit		\$289,00	\$240.00
100-260-035-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-260-040-000	Single Equivalent Dwelling Unit		\$289.00	\$240,00
100-260-041-000	Other Multiple Equivalent Dwelling Units	2	\$578.00	\$480.00
100-260-042-000	Other Multiple Equivalent Dwelling Units	2	\$578.00	\$480.00
100-270-001-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-270-002-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-270-003-000	Single Equivalent Dwelling Unit	d nik timbel se den sambles er belod. 1	\$289.00	\$240.00
100-270-004-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-270-005-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-270-006-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-270-007-000	Single Equivalent Dwelling Unit	rysery, agovat often die bestehrich 1	\$289.00	\$240.00
100-270-008-000	Single Equivalent Dwelling Unit	1	\$289,00	\$240.00
100-270-009-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-270-010-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-270-011-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-270-012-000	Single Equivalent Dwelling Unit	1 7	\$289.00	\$240.00
100-270-013-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-270-014-000	Single Equivalent Dwelling Unit		\$289,00	\$240.00
100-270-015-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-270-016-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-270-017-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-270-018-000	Single Equivalent Dwelling Unit	175	\$289.00	\$240.00
100-270-019-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-270-020-000	Single Equivalent Dwelling Unit	100	\$289.00	\$240.00
100-270-021-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-270-022-000	Single Equivalent Dwelling Unit	1011	\$289.00	\$240.00
100-270-023-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-270-024-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-280-001-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-280-002-000	Single Equivalent Dwelling Unit	i i i i i i i i i i i i i i i i i i i	\$289.00	\$240.00
100-280-003-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-280-004-000	Single Equivalent Dwelling Unit		\$289,00	\$240.00
100-280-005-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-280-006-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-280-007-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-280-008-000	Single Equivalent Dwelling Unit	· · · · · · · · · · · · · · · · · · ·	\$289.00	\$240.00
100-280-009-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-280-010-000	Single Equivalent Dwelling Unit	$\tilde{1}$	\$289.00	\$240.00
100-280-011-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-280-012-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-280-012-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00

100-280-015-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00	Assessor's Parcel Number	Property Classification	Number of EDUs	FY 2023-24 Maximum Special Tax	FY 2023-24 Special Tax Levy
100-280-015-000	100-280-014-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-280-01-000	and the second of the second o		1	\$289.00	\$240.00
100-280-017-000	and the control of th	and the contract of the contra	1	\$289.00	\$240.00
100-280-018-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-019-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-023-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-023-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-025-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-028-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-029-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-030-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-031-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-031-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-033-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-033-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-034-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-035-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-036-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-001-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-005-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-006-000	er a a como anti-como de como de altre a de de contrato de la como de de como de de como de de como de de como	and the state of t	1	\$289.00	\$240.00
100-280-019-000 Single Equivalent Dwelling Unit 1 5289.00 \$240.00 100-280-022-000 Single Equivalent Dwelling Unit 1 5289.00 \$240.00 100-280-023-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-024-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-025-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-025-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-027-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-027-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-028-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-029-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-030-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-030-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-031-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-032-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-033-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-034-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-035-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-001-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-001-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-001-000	A CONTRACTOR OF THE PROPERTY O			\$289.00	\$240.00
100-280-022-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-023-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-025-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-025-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-027-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-027-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-028-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-028-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-028-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-039-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-031-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-031-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-031-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-033-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-033-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-035-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-035-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-036-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-037-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-001-000	 D. William Control of the Control of t	Sang of the Maria and a compared the Maria Maria Maria and Maria	T	\$289.00	\$240.00
1902-280-023-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 \$100-280-025-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 \$240.00 \$100-280-025-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 \$240.00 \$100-280-025-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 \$240.00 \$100-280-027-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 \$240.00 \$100-280-028-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 \$240.00 \$100-280-029-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 \$100-280-039-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 \$100-280-039-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 \$100-280-032-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 \$100-280-032-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 \$100-280-033-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 \$100-280-033-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 \$100-280-035-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 \$100-280-038-000 Other Multiple Equivalent Dwelling Unit 1 \$289.00 \$240.00 \$100-290-001-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 \$100-290-001-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 \$100-290-000-000 Single Equivalent Dwelling Unit 1 \$289	 As a manufacture of the second of the second	and the control of th		\$289.00	\$240.00
100-280-024-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 \$240.00 \$100-280-025-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 \$240.00 \$100-280-027-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 \$240.00 \$100-280-028-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 \$240.00 \$100-280-028-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 \$100-280-028-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 \$100-280-030-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 \$100-280-031-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 \$100-280-031-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 \$100-280-033-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 \$100-280-035-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 \$100-280-035-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 \$100-280-035-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 \$100-280-036-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 \$100-280-038-000 Other Multiple Equivalent Dwelling Unit 1 \$289.00 \$240.00 \$100-290-001-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.	"Market Market State of the Sta	and the state of t	1	The second secon	\$240.00
100-280-025-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 \$240.00 \$100-280-026-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 \$240.00 \$100-280-028-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 \$240.00 \$280-028-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 \$240.00 \$280-028-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 \$240.00 \$280-031-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 \$240.00 \$280-031-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 \$240.00 \$280-031-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 \$240.00 \$280-033-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 \$280-033-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 \$280-034-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 \$280-034-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 \$280-035-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 \$240.00 \$280-030-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 \$280-030-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 \$280-030-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 \$280-030-000 Single Equivalent Dwelling Uni	and the particular and the parti	 And the control of the		\$289,00	\$240.00
100-280-026-000 Single Equivalent Dwelling Unit 1 S289.00 \$240.00 100-280-027-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-029-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-030-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-031-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-036-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-037-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-038-000 Other Multiple Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-001-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-002-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-001-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-002-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-004-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-005-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-006-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-007-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-008-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-009-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-001-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-010-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-	at he as all a favorable to the continuities. A 1980 to	ALCONOMICS AND	1	titing til 17. for a river british meteorialist side statis	\$240.00
100-280-028-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 \$240.00 \$100-280-028-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 \$	and the second control of the second control	The second control of		\$289,00	\$240.00
100-280-028-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-030-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-031-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-031-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-033-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-035-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-035-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-036-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-037-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-037-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-038-000 Other Multiple Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-001-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-001-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-003-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-005-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-005-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-006-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-007-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-007-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-007-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-011-000 Single Equivalent Dwelling Unit 1 \$289	Control and the control of the second	Collins in the Collins of Marketin and Collins in the collins in t	er og selte kinnbære, de distreblik æretereter. 1		\$240,00
100-280-029-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-031-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-031-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-032-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-033-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-033-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-035-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-035-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-035-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-037-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-038-000 Other Multiple Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-038-000 Other Multiple Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-001-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-001-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-002-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-003-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-003-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-005-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-005-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-007-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-010-000 Single Equivalent Dwelling Unit	the same of the course of the property of the course of th			\$289.00	\$240.00
100-280-030-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-032-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-033-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-033-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-034-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-035-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-036-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-037-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-037-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-038-000 Other Multiple Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-001-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-001-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-003-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-003-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-004-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-005-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-005-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-006-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-007-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-008-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-009-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-001-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-010-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-011-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-011-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-015-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-015-000 Single Equivalent Dwelling Unit 1 \$289	the read to be a part of the contribution for the con-		1	\$289.00	\$240.00
100-280-031-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-033-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-033-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-034-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-035-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-035-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-037-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-037-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-037-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-038-000 Other Multiple Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-001-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-002-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-003-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-003-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-005-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-005-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-007-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-008-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-008-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-008-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-009-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-010-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-011-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-011-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-013-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-014-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-015-000 Single Equivalent Dwelling Unit 1 \$289	and the state of t	and an artist of the contract of the second contract of the co		化三氯酚 化二氯化二甲二二甲二甲二甲甲甲甲甲甲甲甲甲甲甲甲甲甲甲甲甲甲甲甲甲甲甲甲甲甲甲	\$240.00
100-280-032-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00	and the second section is a second contract of the second section in the second section is a second section of the second section of the second section is a second section of the section of the second section of the section o	The first of the second of the	1	Substitution of the second section of the second of	\$240.00
100-280-033-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-034-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-035-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-036-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-037-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-038-000 Other Multiple Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-001-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-002-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-002-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-003-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-004-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-005-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-006-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-007-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-008-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-009-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-009-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-010-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-011-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-011-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-013-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-014-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-015-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-016-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-018-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-018-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-	the product of the second of the product of the second of			\$289.00	\$240.00
100-280-034-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-036-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-037-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-037-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-038-000 Other Multiple Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-001-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-002-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-003-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-004-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-004-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-005-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-006-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-007-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-008-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-008-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-010-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-011-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-011-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-013-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-013-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-014-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-015-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-016-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-018-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-018-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-018-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-	the state of the second control of the secon	A CONTRACTOR OF THE PROPERTY O		SERVICE AND ADMINISTRATION OF PARTY	\$240.00
100-280-035-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00	and provide the control of the providence of the			\$289.00	\$240,00
100-280-036-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-037-000 Other Multiple Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-280-038-000 Other Multiple Equivalent Dwelling Units 2 \$578.00 \$480.00 100-290-001-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-002-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-003-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-004-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-005-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-005-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-006-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-007-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-008-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-009-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-010-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-010-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-011-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-012-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-013-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-014-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-015-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-016-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-018-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-018-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-018-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-019-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-019-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00	Special and the second control of the second	and the contract of the contra	1	\$289.00	\$240.00
100-280-037-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00	and the contract of the contra			\$289.00	\$240.00
100-280-038-000 Other Multiple Equivalent Dwelling Units 1 S289.00 S240.00 100-290-001-000 Single Equivalent Dwelling Unit 1 S289.00 S240.00 100-290-002-000 Single Equivalent Dwelling Unit 1 S289.00 S240.00 100-290-003-000 Single Equivalent Dwelling Unit 1 S289.00 S240.00 100-290-004-000 Single Equivalent Dwelling Unit 1 S289.00 S240.00 100-290-005-000 Single Equivalent Dwelling Unit 1 S289.00 S240.00 100-290-006-000 Single Equivalent Dwelling Unit 1 S289.00 S240.00 100-290-007-000 Single Equivalent Dwelling Unit 1 S289.00 S240.00 100-290-008-000 Single Equivalent Dwelling Unit 1 S289.00 S240.00 100-290-008-000 Single Equivalent Dwelling Unit 1 S289.00 S240.00 100-290-009-000 Single Equivalent Dwelling Unit 1 S289.00 S240.00 100-290-009-000 Single Equivalent Dwelling Unit 1 S289.00 S240.00 100-290-010-000 Single Equivalent Dwelling Unit 1 S289.00 S240.00 100-290-011-000 Single Equivalent Dwelling Unit 1 S289.00 S240.00 100-290-012-000 Single Equivalent Dwelling Unit 1 S289.00 S240.00 100-290-013-000 Single Equivalent Dwelling Unit 1 S289.00 S240.00 100-290-013-000 Single Equivalent Dwelling Unit 1 S289.00 S240.00 100-290-014-000 Single Equivalent Dwelling Unit 1 S289.00 S240.00 100-290-016-000 Single Equivalent Dwelling Unit 1 S289.00 S240.00 100-290-016-000 Single Equivalent Dwelling Unit 1 S289.00 S240.00 100-290-016-000 Single Equivalent Dwelling Unit 1 S289.00 S240.00 100-290-017-000 Single Equivalent Dwelling Unit 1 S289.00 S240.00 100-290-018-000 Single Equivalent Dwelling Unit 1 S289.00 S240.00 100-290-018-000 Single Equivalent Dwelling Unit 1 S289.00 S240.00 100-290-019-000 Single Equivalent Dwelling Unit 1 S289.00 S240.00 100-290-019-000 Single Equivalent Dwelling Unit 1 S289.00 S240.00 100-290-019-000 Single Equivalent Dwelling Unit 1 S28	the second contract of		1	the property of the second of the property of the second o	\$240.00
100-290-001-000	the contract of the contract o	and a contract of the contract	2	\$578,00	\$480.00
100-290-002-000	and the second s		1	Age at the contract of the con	\$240.00
100-290-003-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-004-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-005-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-006-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-008-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-009-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-010-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-011-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-012-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-013-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-014-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-015-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00	a company of the contract of t	and a second contract the contract of the cont		\$289.00	\$240.00
100-290-004-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-005-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-006-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-007-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-008-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-009-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-010-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-011-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-012-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-013-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-013-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-014-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-015-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-016-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-017-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-018-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-018-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-019-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-019-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-019-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-020-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-021-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00	and the second of the second o		1	\$289.00	\$240.00
100-290-005-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00	the growing against a commercial growing contract and in the con-			\$289.00	\$240.00
100-290-006-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-007-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-008-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-010-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-011-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-011-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-012-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-013-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-014-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-015-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-016-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-017-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00	The state of the s		1	\$289.00	\$240.00
100-290-007-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-008-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-010-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-011-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-012-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-013-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-014-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-015-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-016-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-017-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-018-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-019-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00	the second contract of	 In the property of the property o		\$289.00	\$240.00
100-290-008-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-009-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-010-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-011-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-012-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-013-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-014-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-015-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-016-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-018-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-019-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-020-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00	and the second of the second o	the service of the control of the co	1	\$289.00	\$240.00
100-290-009-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-010-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-011-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-012-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-013-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-014-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-015-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-016-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-018-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-019-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-020-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-020-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00	the contract of the contract o		1000	management of the contract of	\$240.00
100-290-010-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-011-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-012-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-013-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-014-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-015-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-016-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-018-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-019-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-020-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-020-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-020-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00	and the second of the second o	de la composition de la Transia de Composition de la composition de la composition de la composition de la comp	: 11.1	a della antico della contrata di la	\$240.00
100-290-011-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-012-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-013-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-014-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-015-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-016-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-017-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-019-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-020-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-020-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-020-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-021-000 Exempt 0 \$0.00 \$0.00	and the second s	الإن العلام الإن الإن الأن العالم المناط المناط المناط المناط الأن الأن الأن الأن الأن المناط الإن الإن الإن ا		and the second second and the second second	\$240.00
100-290-012-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-013-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-014-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-015-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-016-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-017-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-018-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-019-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-020-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-020-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-021-000 Exempt 0 \$0.00 \$0.00			1	\$289.00	\$240.00
100-290-013-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-014-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-015-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-016-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-017-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-018-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-019-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-020-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-021-000 Exempt 0 \$0.00 \$0.00	the state of the s	and the programment of the contract of the state of the contract of the contra	1	the control of the control of the control of	region in magazine programme and the contract of the
100-290-014-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-015-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-016-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-017-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-018-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-019-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-020-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-021-000 Exempt 0 \$0.00 \$0.00	and the second control of the contro		1	and the contract of the contra	\$240.00
100-290-015-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-016-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-017-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-018-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-019-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-020-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-021-000 Exempt 0 \$0.00 \$0.00	and the second control of the second control	 In the control of the c	1	grant and the state of the stat	\$240.00
100-290-016-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-017-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-018-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-019-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-020-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-021-000 Exempt 0 \$0.00 \$0.00	the first and the first state of the first state of	- Particle Control (1997) - Particle Contr	1		The second secon
100-290-017-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-018-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-019-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-020-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-021-000 Exempt 0 \$0.00 \$0.00	and the second of the second o	and the contract of the contra	1	A CONTRACTOR OF THE PARTY OF THE PARTY.	\$240.00
100-290-018-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-019-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-020-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-020-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-021-000 Exempt 0 \$0.00 \$0.00	and the second of the second o		1		and the contract of the contra
100-290-019-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-020-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-021-000 Exempt 0 \$0.00 \$0.00	the second control of	the state of the s		and the second of the second o	\$240.00
100-290-020-000 Single Equivalent Dwelling Unit 1 \$289.00 \$240.00 100-290-021-000 Exempt 0 \$0.00 \$0.00	and the second of the second o		1	and the second of the second	and the second of the second o
100-290-021-000 Exempt 0 \$0.00 \$0.00			1	many that the first of the second second	the second control of
100-270-021-000	· · · · · · · · · · · · · · · · · · ·		0	and the second of the first of the	amana a sa sa Caraba da ana ara ara ara ara ara ara ara ara ar
	100-290-022-000	Single Equivalent Dwelling Unit	1		\$240.00

Assessor's Parcel Number	Property Classification	Number of EDUs	FY 2023-24 Maximum Special Tax	FY 2023-24 Special Tax Levy
100-290-023-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-290-024-000	Single Equivalent Dwelling Unit	1	\$289,00	\$240.00
100-300-002-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-300-003-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-300-004-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-300-005-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-300-006-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-300-007-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-300-008-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-300-009-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-300-010-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-300-011-000	Single Equivalent Dwelling Unit	1	\$289,00	\$240.00
100-300-012-000	Single Equivalent Dwelling Unit	1900 - 1100 Series Desired Street Series (1900) 1	\$289.00	\$240.00
100-300-013-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-300-014-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-300-015-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-300-019-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-300-010-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-300-021-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-300-022-000	Single Equivalent Dwelling Unit	1370	\$289.00	\$240.00
100-300-026-000	Other Multiple Equivalent Dwelling Units	2	\$578.00	\$480.00
100-300-020-000	Other Multiple Equivalent Dwelling Units	$\overline{2}$	\$578,00	\$480.00
100-320-001-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-320-001-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-320-003-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-320-003-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240,00
100-320-005-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-320-006-000	Single Equivalent Dwelling Unit	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	\$289.00	\$240.00
100-320-000-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-320-007-000	Single Equivalent Dwelling Unit	u di Miret	\$289.00	\$240.00
100-320-009-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-320-010-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-320-011-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-320-011-000	Single Equivalent Dwelling Unit	i de la companya de	\$289.00	\$240,00
100-320-013-000	Single Equivalent Dwelling Unit	or of the first series. 1	\$289.00	\$240.00
100-320-014-000	Single Equivalent Dwelling Unit	- សស ាធិប៉ុ ល	\$289.00	\$240,00
100-320-015-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-320-015-000	Single Equivalent Dwelling Unit	· · · · · · · · · · · · · · · · · · ·	\$289.00	\$240.00
100-320-017-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-320-017-000	Single Equivalent Dwelling Unit	मार्ग को छ	\$289.00	\$240.00
100-330-001-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-330-001-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-330-002-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-330-004-000	Single Equivalent Dwelling Unit		\$289.00	\$240,00
100-330-003-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
100-330-000-000	Surgia Education Distance Surgia	-		

Assessor's Parcel Number	Property Classification	Number of EDUs	FY 2023-24 Maximum Special Tax	FY 2023-24 Special Tax Levy
100-330-009-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
100-330-009-000	Single Equivalent Dwelling Unit	a	\$289.00	\$240.00
100-330-010-000	Single Equivalent Dwelling Unit		\$289,00	\$240.00
The state of the s	Single Equivalent Dwelling Unit	ang didah di 2012. T	\$289.00	\$240.00
100-330-012-000 100-330-013-000	Single Equivalent Dwelling Unit	er er en	\$289.00	\$240.00
101-080-001-000	Single Equivalent Dwelling Unit	125 Ann 1.436 1	\$289.00	\$240,00
101-080-001-000	Single Equivalent Dwelling Unit	urriurk	\$289.00	\$240.00
101-080-002-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
101-080-004-000	Single Equivalent Dwelling Unit		\$289,00	\$240,00
101-100-007-000	Single Equivalent Dwelling Unit	n amas 1985an dalah. T	\$289.00	\$240.00
101-100-007-000	Single Equivalent Dwelling Unit	anista in a same	\$289.00	\$240.00
101-100-009-000	Single Equivalent Dwelling Unit	ii 1866 il 186 I	\$289.00	\$240.00
101-100-010-000	Single Equivalent Dwelling Unit	oriani	\$289.00	\$240.00
101-100-011-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
101-110-022-000	Optional LPSS Property/Optional Septic Property	$\bar{\mathbf{o}}$	\$0.00	\$0.00
101-110-023-000	Optional LPSS Property/Optional Septic Property	0	\$0.00	\$0.00
101-110-032-000	Single Equivalent Dwelling Unit	13.7	\$289.00	\$240.00
101-110-033-000	Single Equivalent Dwelling Unit	. (*). (), (), (), (). Trans debaardale 1	\$289.00	\$240.00
101-110-034-000	Single Equivalent Dwelling Unit	100	\$289.00	\$240.00
101-120-001-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-120-002-000	Single Equivalent Dwelling Unit	ī	\$289.00	\$240.00
101-120-012-000	Single Equivalent Dwelling Unit	i	\$289.00	\$240.00
101-120-013-000	Single Equivalent Dwelling Unit		\$289,00	\$240.00
101-120-014-000	Single Equivalent Dwelling Unit	1	\$289,00	\$240.00
101-120-015-000	Single Equivalent Dwelling Unit	1111	\$289.00	\$240.00
101-120-016-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-120-019-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-120-020-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-120-030-000	Optional LPSS Property/Optional Septic Property	0	\$0,00	\$0,00
101-120-031-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-120-032-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-120-033-000	Optional LPSS Property/Optional Septic Property	0	\$0.00	\$0.00
101-120-038-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-120-039-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-120-040-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-120-041-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-120-042-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-120-043-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-120-044-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-120-045-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-120-046-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-120-049-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-120-051-000	Single Equivalent Dwelling Unit	1	\$289,00	\$240.00
101-120-052-000	Other Multiple Equivalent Dwelling Units	2	\$578.00	\$480.00
101-130-001-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00

Assessor's Parcel Number	Property Classification	Number of EDUs	FY 2023-24 Maximum Special Tax	FY 2023-24 Special Tax Levy
101-130-004-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-130-005-000	Single Equivalent Dwelling Unit	Maria ya	\$289.00	\$240.00
101-130-006-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-130-007-000	Single Equivalent Dwelling Unit	roeniam	\$289.00	\$240.00
101-130-011-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-130-011-000	Single Equivalent Dwelling Unit	ovenia se	\$289.00	\$240.00
101-130-012-000	Optional LPSS Property/Optional Septic Property	0	\$0.00	\$0.00
101-130-014-000	Single Equivalent Dwelling Unit	ya kii ya k	\$289.00	\$240.00
101-130-014-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-130-015-000	Single Equivalent Dwelling Unit	oras da Šargara d	\$289.00	\$240.00
101-130-017-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-130-017-000	Single Equivalent Dwelling Unit	residence	\$289.00	\$240.00
101-130-019-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-130-019-000	Optional LPSS Property/Optional Septic Property	``` `	\$0.00	\$0.00
101-130-020-000	Single Equivalent Dwelling Unit	h 1	\$289.00	\$240.00
101-130-021-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
101-140-001-000	Single Equivalent Dwelling Unit	desilisi da di barika T	\$289.00	\$240.00
101-140-004-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
101-140-004-000	Single Equivalent Dwelling Unit	oodin bu takaanik 1	\$289.00	\$240.00
101-140-006-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-140-007-000	Single Equivalent Dwelling Unit	andra i i i i i i i i i i i i i i i i i i i	\$289.00	\$240.00
101-140-008-000	Single Equivalent Dwelling Unit		\$289.00	\$240,00
101-140-009-000	Single Equivalent Dwelling Unit	la collègia Bustonico 1	\$289.00	\$240.00
101-140-010-000	Single Equivalent Dwelling Unit	สารองทั้งสาร	\$289,00	\$240.00
101-140-011-000	Single Equivalent Dwelling Unit	organis Aleight III leann di 1	\$289.00	\$240.00
101-140-013-000	Single Equivalent Dwelling Unit	i i	\$289.00	\$240.00
101-140-014-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
101-140-015-000	Single Equivalent Dwelling Unit	u sair u	\$289.00	\$240.00
101-140-015-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-140-017-000	Single Equivalent Dwelling Unit	arans <mark>i</mark> nana	\$289.00	\$240.00
101-140-017-000	Single Equivalent Dwelling Unit	ែក ស្រាមិនសំនេះជាជា 	\$289.00	\$240.00
101-140-019-000	Single Equivalent Dwelling Unit		\$289,00	\$240,00
101-140-020-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-140-021-000	Single Equivalent Dwelling Unit	n garijanan j	\$289,00	\$240.00
101-140-021-000	Single Equivalent Dwelling Unit	viil de de libralité d' 1	\$289.00	\$240.00
101-140-022-000	Single Equivalent Dwelling Unit	okazan <mark>i</mark> sangan	\$289.00	\$240.00
101-140-023-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
the second of th	Single Equivalent Dwelling Unit	an mingra	\$289.00	\$240,00
101-140-025-000	Single Equivalent Dwelling Unit	erin i diliku Shisi. 1	\$289.00	\$240.00
101-140-026-000	Single Equivalent Dwelling Unit	ត្រង់ជាំប្រ	\$289.00	\$240.00
101-140-027-000 101-140-028-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-140-028-000	Duplex	or of the	\$578.00	\$480.00
101-150-023-000	Optional LPSS Property/Optional Septic Property	.	\$0.00	\$0.00
101-150-024-000	Optional LPSS Property/Optional Septic Property	<u> </u>	\$0.00	\$0.00
101-150-025-000	Optional LPSS Property/Optional Septic Property	0	\$0.00	\$0.00
101-150-020-000	Optional Li bo Froperty/Optional Septic Froperty	Ü	ΨΟΙΟΟ	W0100

Assessor's Parcel Number	Property Classification	Number of EDUs	FY 2023-24 Maximum Special Tax	FY 2023-24 Special Tax Levy
101-160-001-000	Optional LPSS Property/Optional Septic Property	0	\$0,00	\$0.00
101-160-002-510	Optional LPSS Property/Optional Septic Property	0	\$0.00	\$0.00
101-160-002-520	Optional LPSS Property/Optional Septic Property	0	\$0.00	\$0.00
101-160-003-000	Optional LPSS Property/Optional Septic Property	0	\$0.00	\$0.00
101-160-004-000	Optional LPSS Property/Optional Septic Property	0	\$0.00	\$0.00
101-160-005-000	Optional LPSS Property/Optional Septic Property	0	\$0.00	\$0.00
101-160-006-000	Duplex	2	\$578.00	\$480.00
101-160-007-000	Optional LPSS Property/Optional Septic Property	0	\$0,00	\$0.00
101-160-008-000	Optional LPSS Property/Optional Septic Property	0	\$0,00	\$0,00
101-160-009-000	Optional LPSS Property/Optional Septic Property	0	\$0.00	\$0.00
101-160-010-000	Optional LPSS Property/Optional Septic Property	0	\$0.00	\$0.00
101-160-011-000	Optional LPSS Property/Optional Septic Property	0	\$0.00	\$0.00
101-160-012-000	Optional LPSS Property/Optional Septic Property	0	\$0.00	\$0.00
101-160-013-000	Duplex	2	\$578.00	\$480.00
101-160-014-000	Duplex	2	\$578.00	\$480.00
101-160-015-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-160-016-000	Optional LPSS Property/Optional Septic Property	0	\$0.00	\$0.00
101-160-017-000	Optional LPSS Property/Optional Septic Property	0	\$0.00	\$0,00
101-160-018-000	Optional LPSS Property/Optional Septic Property	0	\$0.00	\$0.00
101-160-019-000	Optional LPSS Property/Optional Septic Property	0	\$0.00	\$0.00
101-160-020-000	Optional LPSS Property/Optional Septic Property	0	\$0.00	\$0.00
101-160-021-000	Optional LPSS Property/Optional Septic Property	0	\$0.00	\$0.00
101-160-022-000	Optional LPSS Property/Optional Septic Property	0	\$0.00	\$0.00
101-160-023-000	Optional LPSS Property/Optional Septic Property	0	\$0.00	\$0,00
101-160-024-000	Optional LPSS Property/Optional Septic Property	0	\$0.00	\$0.00
101-160-025-000	Optional LPSS Property/Optional Septic Property	0	\$0.00	\$0.00
101-160-026-000	Optional LPSS Property/Optional Septic Property	0	\$0.00	\$0.00
101-160-027-000	Optional LPSS Property/Optional Septic Property	0	\$0.00	\$0.00
101-160-028-000	Optional LPSS Property/Optional Septic Property	0	\$0.00	\$0.00
101-160-029-000	Optional LPSS Property/Optional Septic Property	0	\$0.00	\$0.00
101-160-030-000	Single Equivalent Dwelling Unit	ī	\$289.00	\$240.00
101-160-031-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240,00
101-180-010-000	Single Equivalent Dwelling Unit		\$289,00	\$240.00
101-180-011-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-180-012-000	Single Equivalent Dwelling Unit		\$289,00	\$240.00
101-180-013-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-180-022-000	Single Equivalent Dwelling Unit	\mathbf{i}	\$289.00	\$240.00
101-180-023-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-180-024-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-180-025-000	Single Equivalent Dwelling Unit	1	\$289,00	\$240,00
101-180-026-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-180-027-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-180-028-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-180-029-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-180-030-000	Single Equivalent Dwelling Unit	1	\$289,00	\$240.00

Assessor's Parcel Number	Property Classification	Number of EDUs	FY 2023-24 Maximum Special Tax	FY 2023-24 Special Tax Levy
101-180-031-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-180-032-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-180-033-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-180-034-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
101-180-037-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-180-038-000	Single Equivalent Dwelling Unit		\$289,00	\$240.00
101-180-039-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-180-040-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
101-180-041-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-180-042-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
101-180-043-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-180-044-000	Single Equivalent Dwelling Unit	1 7	\$289,00	\$240,00
101-180-048-000	Other Multiple Equivalent Dwelling Units	.90182-7.932-769-260-2	\$578.00	\$480.00
101-250-001-000	Single Equivalent Dwelling Unit		\$289,00	\$240,00
101-250-002-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-250-002-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
101-250-003-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-250-005-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
101-250-005-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-250-007-000	Single Equivalent Dwelling Unit	u sarri da d	\$289.00	\$240.00
101-250-007-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
101-250-009-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
101-250-010-000	Single Equivalent Dwelling Unit	indi de da Como de distri	\$289.00	\$240.00
101-250-011-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
101-250-012-000	Single Equivalent Dwelling Unit	T	\$289.00	\$240.00
101-250-013-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
101-250-014-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
101-250-015-000	Single Equivalent Dwelling Unit	ta ta a faire.	\$289.00	\$240.00
101-250-015-000	Single Equivalent Dwelling Unit	ilkus vari illus friterastus. 1	\$289.00	\$240.00
101-250-017-000	Single Equivalent Dwelling Unit	o o o o o a figura	\$289.00	\$240.00
101-250-017-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-250-019-000	Single Equivalent Dwelling Unit	r sameni <mark>ja 7</mark> 9.	\$289.00	\$240.00
101-250-020-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
101-250-021-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
101-250-021-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-250-022-000	Single Equivalent Dwelling Unit	នាការ ជ ៀ នបាន	\$289.00	\$240.00
and the control of th	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-250-024-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
101-250-025-000 101-250-026-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
the contract of the contract o	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-250-029-000 101-250-031-000	Other Multiple Equivalent Dwelling Units		\$578.00	\$480.00
101-260-001-000	Single Equivalent Dwelling Unit	1	\$289,00	\$240.00
The state of the s	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-260-002-000	Single Equivalent Dwelling Unit	1 1	\$289.00	\$240.00
101-260-003-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
101-260-004-000	Single Equivalent Dweiting Out	r	\$207.00	72.0.00

Assessor's Parcel Number	Property Classification	Number of EDUs	FY 2023-24 Maximum Special Tax	FY 2023-24 Special Tax Levy
101-260-007-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
101-260-008-000	Single Equivalent Dwelling Unit	52, 30,898, 41 M A A A A A A A A A A A A A A A A A A	\$289.00	\$240.00
101-260-009-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
101-260-012-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-260-013-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
101-260-014-000	Single Equivalent Dwelling Unit	a, a, et estreale est est a alta est de la	\$289.00	\$240.00
101-260-015-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
101-260-016-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-260-017-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
101-260-018-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-260-019-000	Single Equivalent Dwelling Unit		\$289,00	\$240.00
101-260-020-000	Single Equivalent Dwelling Unit	in etak stania kalabas in a aria saasaa. 1	\$289.00	\$240.00
101-260-021-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
101-260-022-000	Single Equivalent Dwelling Unit	essenti (narandarra vye si nemesta instrumenta 1	\$289.00	\$240.00
101-260-023-000	Single Equivalent Dwelling Unit		\$289,00	\$240.00
101-260-024-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-260-029-000	Exempt	0	\$0.00	\$0.00
101-260-030-000	Single Equivalent Dwelling Unit]	\$289.00	\$240.00
101-260-033-000	Single Equivalent Dwelling Unit		\$289.00	\$240,00
101-260-034-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-260-035-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
101-260-036-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-260-037-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
101-260-038-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-260-039-000	Single Equivalent Dwelling Unit	i	\$289.00	\$240,00
101-260-040-000	Other Multiple Equivalent Dwelling Units	2	\$578.00	\$480.00
101-270-001-000	Single Equivalent Dwelling Unit		\$289,00	\$240.00
101-270-002-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-270-003-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
101-270-004-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-270-005-000	Single Equivalent Dwelling Unit		\$289.00	\$240,00
101-270-006-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-270-007-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-270-008-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-270-009-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-270-010-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-270-011-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
101-270-012-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-270-013-000	Exempt	0	\$0,00	\$0,00
101-270-016-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-270-017-000	Single Equivalent Dwelling Unit	1	\$289,00	\$240.00
101-270-018-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-270-019-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-270-020-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-270-021-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00

Assessor's Parcel Number	Property Classification	Number of EDUs	FY 2023-24 Maximum Special Tax	FY 2023-24 Special Tax Levy
101-270-022-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-270-023-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
101-270-024-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-270-025-000	Single Equivalent Dwelling Unit	7 21	\$289,00	\$240.00
101-270-026-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-270-027-000	Single Equivalent Dwelling Unit	I	\$289.00	\$240.00
101-270-028-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-270-029-000	Single Equivalent Dwelling Unit		\$289,00	\$240.00
101-270-030-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-270-031-000	Single Equivalent Dwelling Unit		\$289,00	\$240.00
101-270-032-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-270-034-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
101-270-035-000	Single Equivalent Dwelling Unit	. 382. Pilo di rassili anal-di alci Pri 1	\$289.00	\$240.00
101-270-036-000	Single Equivalent Dwelling Unit	T	\$289.00	\$240.00
101-270-037-000	Other Multiple Equivalent Dwelling Units		\$578.00	\$480.00
101-280-001-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
101-280-002-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-280-003-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
101-280-004-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-280-005-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
101-280-006-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-280-008-000	Single Equivalent Dwelling Unit		\$289,00	\$240.00
101-280-009-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-280-010-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
101-280-011-000	Optional LPSS Property/Optional Septic Property	0	\$0.00	\$0.00
101-280-012-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
101-280-013-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-280-014-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
101-280-015-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-280-016-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-280-017-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-280-018-000	Single Equivalent Dwelling Unit	Ti N	\$289.00	\$240,00
101-280-019-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-280-020-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-280-021-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-280-022-000	Single Equivalent Dwelling Unit	1	\$289,00	\$240.00
101-280-023-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240,00
101-280-024-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-280-025-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-280-026-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-280-027-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-280-028-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-280-029-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-280-030-000	Single Equivalent Dwelling Unit	.1	\$289.00	\$240.00
101-280-031-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00

Assessor's Parcel Number	Property Classification	Number of EDUs	FY 2023-24 Maximum Special Tax	FY 2023-24 Special Tax Levy
101-280-032-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
101-280-033-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-280-034-000	Single Equivalent Dwelling Unit		\$289,00	\$240.00
101-280-035-000	Single Equivalent Dweiling Unit	1	\$289.00	\$240.00
101-280-036-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
101-280-037-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-280-038-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
101-280-039-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-280-040-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-280-041-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-280-042-000	Single Equivalent Dwelling Unit	1, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1	\$289.00	\$240,00
101-280-043-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-280-044-000	Single Equivalent Dwelling Unit		\$289.00	\$240,00
101-280-045-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-280-046-000	Single Equivalent Dwelling Unit		\$289.00	\$240,00
101-280-047-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-280-048-000	Single Equivalent Dwelling Unit	1	\$289,00	\$240.00
101-290-001-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-290-002-000	Single Equivalent Dwelling Unit		\$289,00	\$240.00
101-290-003-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-290-004-000	Single Equivalent Dwelling Unit	1	\$289,00	\$240.00
101-290-005-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-290-008-000	Single Equivalent Dwelling Unit	1	\$289,00	\$240,00
101-290-009-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-290-010-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
101-290-011-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-290-012-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-290-013-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-290-014-000	Single Equivalent Dwelling Unit	1	\$289,00	\$240.00
101-290-015-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-290-018-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-290-019-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240,00
101-290-020-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
101-290-021-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-290-022-000	Single Equivalent Dwelling Unit	1	\$289,00	\$240.00
101-290-023-000	Single Equivalent Dwelling Unit	1	\$289,00	\$240.00
101-290-024-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-290-025-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-290-026-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240,00
101-290-027-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-290-028-000	Single Equivalent Dwelling Unit	1	\$289,00	\$240.00
101-290-029-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-290-030-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-290-031-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-290-032-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00

Assessor's Parcel Number	Property Classification	Number of EDUs	FY 2023-24 Maximum Special Tax	FY 2023-24 Special Tax Levy
101-290-033-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-290-034-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
101-290-035-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-290-036-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
101-290-038-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-290-039-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
101-290-040-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-290-041-000	Single Equivalent Dwelling Unit	i	\$289.00	\$240.00
101-290-042-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
101-290-043-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-010-007-000	Single Equivalent Dwelling Unit	laid alain a alaman la lain al ma 1	\$289.00	\$240.00
102-010-008-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-010-009-000	Single Equivalent Dwelling Unit	oud welds and the said of 2019.	\$289.00	\$240.00
102-010-010-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-010-011-000	Taxable Property Owner Association Property	ili da saya ma Masa a sa a sa a sa a sa a sa a sa	\$289.00	\$240.00
102-020-004-000	Single Equivalent Dwelling Unit		\$289.00	\$240,00
102-020-005-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-020-006-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-020-007-000	Triplex	an parting to Pictine nith. 3	\$867.00	\$720.00
102-020-008-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-020-009-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-020-010-000	Single Equivalent Dwelling Unit	i = i	\$289.00	\$240.00
102-020-011-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-020-012-000	Single Equivalent Dwelling Unit	ા જોવા જે	\$289.00	\$240.00
102-020-013-000	Single Equivalent Dwelling Unit	. The analytical sector of the	\$289.00	\$240,00
102-020-014-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-020-015-000	Single Equivalent Dwelling Unit	1969 - 41 - 1961 - 1974 - 1984 - 1984 - 1984 - 1984 - 1984 - 1984 - 1984 - 1984 - 1984 - 1984 - 1984 - 1984 - 1	\$289.00	\$240.00
102-020-018-000	Single Equivalent Dwelling Unit	17.50	\$289,00	\$240.00
102-020-019-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-020-019-000	Single Equivalent Dwelling Unit	o y light	\$289.00	\$240.00
102-020-022-000	Other Multiple Equivalent Dwelling Units	3	\$867.00	\$720.00
102-020-023-000	Other Multiple Equivalent Dwelling Units	2	\$578.00	\$480.00
102-030-001-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-030-002-000	Duplex	$\frac{1}{2}$	\$578.00	\$480,00
102-030-003-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-030-004-000	Single Equivalent Dwelling Unit	un e eferiale	\$289,00	\$240.00
102-030-005-000	Single Equivalent Dwelling Unit	ada a restriction and dis- 1	\$289.00	\$240.00
102-030-005-000	Single Equivalent Dwelling Unit		\$289,00	\$240,00
102-030-007-000	Single Equivalent Dwelling Unit	i kalendila da ilde 1	\$289.00	\$240.00
102-030-008-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-030-009-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-030-009-000	Single Equivalent Dwelling Unit	an er ar 🛊 ammer	\$289.00	\$240.00
102-030-010-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-030-011-000	Single Equivalent Dwelling Unit	· · · · · · · · · · · · · · · · · · ·	\$289.00	\$240.00
102-030-012-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
			*	

Assessor's Parcel Number	Property Classification	Number of EDUs	FY 2023-24 Maximum Special Tax	FY 2023-24 Special Tax Levy
102-030-014-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-030-015-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-030-016-000	Single Equivalent Dwelling Unit		\$289,00	\$240.00
102-030-017-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-030-018-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-030-019-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-040-001-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-040-002-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-040-003-000	Single Equivalent Dwelling Unit		\$289.00	\$240,00
102-040-004-000	Duplex	2	\$578.00	\$480.00
102-040-005-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-040-006-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240,00
102-040-007-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-040-008-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-040-009-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-040-010-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-040-011-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-040-012-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-040-013-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-040-014-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-040-015-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-040-016-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-051-001-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240,00
102-051-002-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240,00
102-051-003-000	Duplex	2	\$578.00	\$480.00
102-051-004-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-051-005-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-051-006-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-051-007-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-051-008-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-051-009-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-051-010-000	Single Equivalent Dwelling Unit	1	\$289,00	\$240,00
102-051-011-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-051-012-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240,00
102-051-013-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-051-014-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240,00
102-051-015-000	Single Equivalent Dwelling Unit	_1	\$289,00	\$240,00
102-052-003-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-052-004-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-052-005-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-052-006-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-052-007-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-052-008-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-052-009-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-052-010-000	Single Equivalent Dwelling Unit	.1	\$289.00	\$240.00

Assessor's Parcel Number	Property Classification	Number of EDUs	FY 2023-24 Maximum Special Tax	FY 2023-24 Special Tax Levy
102-052-011-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-052-012-000	Single Equivalent Dwelling Unit	100	\$289.00	\$240.00
102-052-013-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-052-014-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-052-015-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-052-016-000	Single Equivalent Dwelling Unit	i i	\$289.00	\$240.00
102-052-017-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-052-018-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-052-019-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-052-020-000	Other Multiple Equivalent Dwelling Units	2 7	\$578.00	\$480.00
102-053-001-000	Single Equivalent Dwelling Unit	1	\$289,00	\$240.00
102-053-002-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-053-003-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-053-004-000	Single Equivalent Dwelling Unit	i i i i i i i i i i i i i i i i i i i	\$289.00	\$240.00
102-053-005-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-053-006-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-053-007-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-053-008-000	Single Equivalent Dwelling Unit	THE THE	\$289.00	\$240.00
102-053-009-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-053-012-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-053-013-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-053-014-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-053-015-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-053-016-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240,00
102-053-017-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-053-018-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-053-019-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-053-020-000	Single Equivalent Dwelling Unit	1	\$289,00	\$240.00
102-054-001-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-054-002-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-054-003-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-054-004-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-054-005-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-054-006-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240,00
102-054-007-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-054-008-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-054-010-000	Single Equivalent Dwelling Unit	1	\$289,00	\$240,00
102-054-011-000	Single Equivalent Dwelling Unit	. 1	\$289.00	\$240.00
102-054-012-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-054-013-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-054-014-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-054-015-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-054-016-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-054-017-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-054-018-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00

Assessor's Parcel Number	Property Classification	Number of EDUs	FY 2023-24 Maximum Special Tax	FY 2023-24 Special Tax Levy
102-054-019-000	Single Equivalent Dwelling Unit		\$289.00	\$240,00
102-054-020-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-055-001-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-055-002-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240,00
102-055-003-000	Single Equivalent Dwelling Unit	1	\$289,00	\$240.00
102-055-004-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-055-005-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-055-006-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-055-007-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-055-009-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-055-012-000	Single Equivalent Dwelling Unit		\$289,00	\$240.00
102-055-015-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-055-016-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-055-020-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-055-021-000	Single Equivalent Dwelling Unit		\$289.00	\$240,00
102-055-022-000	Other Multiple Equivalent Dwelling Units	2	\$578.00	\$480.00
102-055-023-000	Other Multiple Equivalent Dwelling Units	2	\$578.00	\$480,00
102-060-003-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-060-004-000	Single Equivalent Dwelling Unit		\$289.00	\$240,00
102-060-005-000	Single Equivalent Dwelling Unit	1	\$289,00	\$240.00
102-060-006-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-060-007-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-060-008-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240,00
102-060-011-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-060-012-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-060-013-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-060-016-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-060-017-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-060-018-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-060-019-000	Single Equivalent Dweiling Unit	1	\$289.00	\$240.00
102-060-020-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-060-021-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-060-022-000	Single Equivalent Dwelling Unit	11	\$289.00	\$240.00
102-060-023-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-060-024-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-060-025-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-060-026-000	Other Multiple Equivalent Dwelling Units	2	\$578.00	\$480,00
102-070-010-000	Exempt	0	\$0.00	\$0.00
102-070-011-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-070-012-000	Taxable Property Owner Association Property	2	\$578.00	\$480.00
102-070-013-000	Exempt	0	\$0.00	\$0.00
102-080-001-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-080-003-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240,00
102-080-004-000	Duplex	2	\$578.00	\$480.00
102-080-005-000	Single Equivalent Dwelling Unit	1	\$289,00	\$240.00

Assessor's Parcel Number	Property Classification	Number of EDUs	FY 2023-24 Maximum Special Tax	FY 2023-24 Special Tax Levy
102-080-006-000	Single Equivalent Dwelling Unit	` 1	\$289.00	\$240.00
102-080-007-000	Single Equivalent Dwelling Unit		\$289,00	\$240.00
102-080-008-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-080-009-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-080-010-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-080-011-000	Single Equivalent Dwelling Unit	TEVAVĒTE	\$289.00	\$240,00
102-080-012-000	Single Equivalent Dwelling Unit	ikincinisahe atkan isa 1	\$289.00	\$240.00
102-080-013-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-080-014-000	Single Equivalent Dwelling Unit	u gund i saundi 71 gafdalar 1	\$289.00	\$240.00
102-080-015-000	Single Equivalent Dwelling Unit	roundining.	\$289,00	\$240.00
102-080-016-000	Single Equivalent Dwelling Unit	Bettern is dae de berêtekt niddet and 1	\$289,00	\$240,00
102-080-017-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-080-018-000	Single Equivalent Dwelling Unit	altinit luai takii kad	\$289.00	\$240.00
102-080-019-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-080-021-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-080-021-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-080-023-000	Single Equivalent Dwelling Unit	ndaga gagban bir fixtiki saki 1	\$289.00	\$240.00
102-080-023-000	Single Equivalent Dwelling Unit	ranjak j a ees	\$289.00	\$240.00
102-080-025-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-080-025-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-080-027-000	Duplex	100 4 - 10 #is/ ak is 2	\$578.00	\$480.00
102-080-028-000	Duplex	อาเกาะสร้ายสา	\$578.00	\$480.00
102-080-029-000	Single Equivalent Dwelling Unit	ir	\$289.00	\$240.00
102-080-030-000	Single Equivalent Dwelling Unit	orano e si in esta e de la compansión de l La compansión de la compa	\$289.00	\$240.00
102-080-031-000	Single Equivalent Dwelling Unit]	\$289.00	\$240,00
102-080-031-000	Single Equivalent Dwelling Unit	r meire	\$289.00	\$240,00
102-080-032-000	Single Equivalent Dwelling Unit	. O carliti i dile <mark>t</mark> i di 2000	\$289.00	\$240.00
102-080-034-000	Single Equivalent Dwelling Unit		\$289,00	\$240,00
102-080-035-000	Single Equivalent Dwelling Unit	ninauirusirusi t uikakusi 1	\$289.00	\$240.00
102-080-035-000	Single Equivalent Dwelling Unit	ng marking ga	\$289.00	\$240.00
102-080-037-000	Single Equivalent Dwelling Unit	produce linear producer and	\$289.00	\$240.00
102-080-037-000	Single Equivalent Dwelling Unit	er mor pågere	\$289.00	\$240.00
102-080-039-000	Single Equivalent Dwelling Unit	11.55 (i.e. 11.11.11.11.11.11.11.11.11.11.11.11.11	\$289.00	\$240.00
and the second control of the second control		arawa an mijan <mark>i</mark> an, mwanj	\$289.00	\$240.00
102-080-040-000	Single Equivalent Dwelling Unit		and all the control of the control o	\$0.00
102-080-041-000	Exempt	i mir kalanda 🕌 kalanda	\$0.00	\$240,00
102-080-042-000	Single Equivalent Dwelling Unit	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	\$289.00	
102-080-045-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-080-046-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-080-047-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-080-050-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-080-051-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-080-052-000	Duplex	2	\$578.00	\$480,00
102-080-055-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-080-056-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-080-057-000	Exempt	0	\$0.00	\$0.00

Assessor's Parcel Number	Property Classification	Number of EDUs	FY 2023-24 Maximum Special Tax	FY 2023-24 Special Tax Levy
102-080-059-000	Duplex	2	\$578.00	\$480.00
102-080-060-000	Single Equivalent Dwelling Unit	1	\$289,00	\$240.00
102-080-064-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-090-001-000	Single Equivalent Dwelling Unit	l	\$289.00	\$240.00
102-090-002-000	Single Equivalent Dwelling Unit		\$289,00	\$240,00
102-090-003-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-090-004-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-090-005-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-090-006-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-090-007-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-090-008-000	Single Equivalent Dwelling Unit		\$289,00	\$240.00
102-090-009-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-090-010-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-090-011-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-090-012-000	Single Equivalent Dwelling Unit		\$289,00	\$240.00
102-090-013-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-090-014-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-090-015-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-090-016-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-090-017-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-090-018-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-090-019-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-090-020-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240,00
102-090-021-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-090-022-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-090-023-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-090-024-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-090-025-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-090-026-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-090-027-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-090-028-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-090-029-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-090-030-000	Duplex	2	\$578.00	\$480.00
102-090-031-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-090-032-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-090-033-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-090-034-000	Single Equivalent Dwelling Unit	1	\$289,00	\$240.00
102-090-035-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-090-036-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-090-037-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-090-038-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-090-039-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-090-040-000	Single Equivalent Dwelling Unit	. 1	\$289.00	\$240.00
102-090-041-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-090-042-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240,00

Assessor's Parcel Number	Property Classification	Number of EDUs	FY 2023-24 Maximum Special Tax	FY 2023-24 Special Tax Levy
102-090-043-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-090-044-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-090-045-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-090-046-000	Single Equivalent Dwelling Unit		\$289.00	\$240,00
102-090-047-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-090-048-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-090-049-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-090-052-000	Single Equivalent Dwelling Unit		\$289,00	\$240.00
102-090-053-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-090-054-000	Single Equivalent Dwelling Unit		\$289,00	\$240.00
102-090-055-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-090-056-000	Single Equivalent Dwelling Unit	11111	\$289,00	\$240.00
102-090-059-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-090-060-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-090-061-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-090-062-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-090-063-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-090-064-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-090-065-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-090-066-000	Exempt	0	\$0,00	\$0.00
102-090-067-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-090-068-000	Other Multiple Equivalent Dwelling Units	2	\$578.00	\$480.00
102-100-001-000	Single Equivalent Dwelling Unit	1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 -	\$289.00	\$240,00
102-100-002-000	Single Equivalent Dwelling Unit	1	\$289,00	\$240.00
102-100-003-000	Single Equivalent Dwelling Unit	1 -	\$289.00	\$240.00
102-100-004-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-100-005-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-100-006-000	Single Equivalent Dwelling Unit		\$289,00	\$240.00
102-100-007-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-100-008-000	Single Equivalent Dwelling Unit	1	\$289,00	\$240.00
102-100-011-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-100-012-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-100-013-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-100-014-000	Single Equivalent Dwelling Unit	i = i	\$289.00	\$240,00
102-100-015-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240,00
102-100-016-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-100-020-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-100-021-000	Single Equivalent Dwelling Unit	1	\$289,00	\$240.00
102-100-022-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-100-023-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-100-024-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-100-025-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-100-030-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-100-031-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-100-032-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00

Assessor's Parcel Number	Property Classification	Number of EDUs	FY 2023-24 Maximum Special Tax	FY 2023-24 Special Tax Levy
102-100-033-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-100-034-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-100-035-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-100-036-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-100-037-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-100-040-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-100-041-000	Single Equivalent Dwelling Unit		\$289,00	\$240.00
102-100-042-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-100-043-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-100-044-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-100-048-000	Single Equivalent Dwelling Unit		\$289.00	\$240,00
102-100-049-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-100-050-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-100-051-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-100-052-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-100-053-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-100-054-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240,00
102-100-055-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-100-056-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-100-057-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-100-058-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-100-059-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-100-060-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-100-061-000	Single Equivalent Dwelling Unit	1	\$289,00	\$240.00
102-100-064-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-100-065-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-100-068-000	Single Equivalent Dwelling Unit	i	\$289.00	\$240.00
102-100-070-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-100-071-000	Other Multiple Equivalent Dwelling Units	2	\$578.00	\$480.00
102-100-072-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-100-073-000	Taxable Property Owner Association Property	1	\$289.00	\$240.00
102-100-074-000	Other Multiple Equivalent Dwelling Units	2	\$578.00	\$480.00
102-100-075-000	Other Multiple Equivalent Dwelling Units	2	\$578,00	\$480.00
102-110-001-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-110-002-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-110-003-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-110-004-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-110-005-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-110-006-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-110-007-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-110-008-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-110-009-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-110-010-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-110-011-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-110-012-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240,00

Assessor's Parcel Number	Property Classification	Number of EDUs	FY 2023-24 Maximum Special Tax	FY 2023-24 Special Tax Levy
T at cer ivulibles		01 220 03		
102-110-016-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-110-017-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-110-018-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-110-019-000	Single Equivalent Dwelling Unit		\$289.00	\$240,00
102-110-020-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240,00
102-110-021-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-110-022-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-110-025-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-110-026-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-110-027-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-110-030-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-110-031-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-110-032-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-110-037-000	Single Equivalent Dwelling Unit	1.41	\$ 289,00	\$240.00
102-110-040-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-110-041-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-110-042-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-110-043-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-110-044-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-110-045-000	Single Equivalent Dwelling Unit	1.55	\$289.00	\$240.00
102-110-046-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-110-047-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-110-048-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-110-049-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-110-050-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-110-055-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-110-057-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-110-060-000	Other Multiple Equivalent Dwelling Units	2	\$578.00	\$480.00
102-110-061-000	Other Multiple Equivalent Dwelling Units	2	\$578.00	\$480.00
102-110-062-000	Other Multiple Equivalent Dwelling Units	2	\$578.00	\$480.00
102-110-063-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-110-064-000	Other Multiple Equivalent Dwelling Units	2	\$578,00	\$480.00
102-110-065-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-110-066-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-110-067-000	Other Multiple Equivalent Dwelling Units	2	\$578.00	\$480.00
102-120-001-000	Single Equivalent Dwelling Unit	1	\$289,00	\$240.00
102-120-002-000	Single Equivalent Dwelling Unit	1	\$289,00	\$240.00
102-120-003-000	Single Equivalent Dwelling Unit	I	\$289.00	\$240.00
102-120-008-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-120-009-000	Exempt	Ö	\$0.00	\$0.00
102-120-018-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-120-019-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-120-024-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-120-025-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-120-026-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00

Assessor's Parcel Number	Property Classification	Number of EDUs	FY 2023-24 Maximum Special Tax	FY 2023-24 Special Tax Levy
	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-120-027-000 102-120-028-000	Single Equivalent Dwelling Unit	1 1 1 1 1 1 1 1 1 1	\$289.00	\$240.00
102-120-028-000	Single Equivalent Dwelling Unit	enge versijen vijge	\$289.00 \$289.00	\$240.00
102-120-029-000	Single Equivalent Dwelling Unit	Ziinida÷akatai 1	\$289.00	\$240.00
102-120-031-000	Single Equivalent Dwelling Unit		\$289,00	\$240.00 \$240.00
and the second s	Single Equivalent Dwelling Unit	Nelso shural And Live kil 1	\$289.00	\$240.00
102-120-032-000 102-120-033-000	Single Equivalent Dwelling Unit		\$289.00 \$289.00	\$240.00 \$240.00
A Common March State and March State of State of the State of State of the State of	Single Equivalent Dwelling Unit	id ik silab id ^a s kalabili. T	\$289.00	\$240.00
102-120-034-000 102-120-035-000	Single Equivalent Dwelling Unit		\$289.00	\$240,00
102-120-035-000	of C. Charles and the Control of the		\$289.00	\$240.00
102-120-036-000	Single Equivalent Dwelling Unit Single Equivalent Dwelling Unit		\$289.00	\$240.00
 To the content of the c	in the carties. What is a califfic even to Taracteria Personalisade ten com a What is with the Committee of the	//////////////////////////////////////	\$289.00	\$240.00
102-120-038-000	Single Equivalent Dwelling Unit		\$289,00	\$240.00
102-120-039-000 102-120-040-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
Approximate and the first of the second section of the section of the second section of the section of the second section of the section of t	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-120-041-000	Single Equivalent Dwelling Unit Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-120-042-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-120-043-000	න දෙදෙනා පත්තේ කාර්ත මෙන වීම් දෙනකුරුම් දක මාධා දෙදෙනාක පත්තර සදහාමී නම් දෙනත් එම්ම ජීමාව මාදහ	1	\$289.00 \$289.00	\$240.00
102-120-044-000	Single Equivalent Dwelling Unit		\$289.00 \$289.00	\$240.00
102-120-045-000 102-120-046-000	Single Equivalent Dwelling Unit	(1. kg - 1. kg ±1.438 1.12 1	\$289.00	\$240.00
102-120-046-000	Single Equivalent Dwelling Unit Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-120-047-000	Single Equivalent Dwelling Unit Single Equivalent Dwelling Unit	Takata Mehabbasa T	\$289.00	\$240.00
102-120-049-000	in and the same at a regard and force the regarding means of the contract of the contract and the contract of		\$289.00	\$240.00
102-120-049-000	Single Equivalent Dwelling Unit Single Equivalent Dwelling Unit	in dan ilahabada. T	\$289.00	\$240.00
102-120-053-000	Single Equivalent Dwelling Unit Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-120-055-000	Other Multiple Equivalent Dwelling Units		\$578.00	\$480.00
المروانية والمستقوم بالمستفاد والمتاريق	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-120-058-000	Taxable Property Owner Association Property	and de arkindidê 1	\$289.00	\$240.00
102-120-060-000 102-120-061-000	skur danga kapanan mengapangkan Majamba Talah dan dialah dan dan lama mengalip per, pagan bengGilangka Tilah d	, i	\$0.00	\$0.00
102-120-061-000	Exempt Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-120-062-000	Other Multiple Equivalent Dwelling Units	1	\$578.00	\$480.00
102-120-064-000	Other Multiple Equivalent Dwelling Units Other Multiple Equivalent Dwelling Units	a and Arabida and the A	\$578.00	\$480.00
102-120-004-000	Single Equivalent Dwelling Unit	arraga <mark>f</mark> i wasa	\$289.00	\$240.00
	Single Equivalent Dwelling Unit	omenimi v 📩 i se se se T	\$289.00	\$240.00
102-130-004-000 102-130-005-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-130-005-000	Single Equivalent Dwelling Unit Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-130-007-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-130-007-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
the second contract of	Single Equivalent Dwelling Unit	para a si <mark>t</mark> a a siya	\$289.00	\$240.00
102-130-009-000 102-130-012-000	Single Equivalent Dwelling Unit	1 1	\$289.00	\$240.00
102-130-012-000	Single Equivalent Dwelling Unit Single Equivalent Dwelling Unit	· , · · · · · · · · · · · · · · · · · ·	\$289.00	\$240.00
102-130-013-000	Single Equivalent Dwelling Unit	· · · · · · · · · · · · · · · · · · ·	\$289.00	\$240.00
102-130-014-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-130-015-000	Single Equivalent Dwelling Unit Single Equivalent Dwelling Unit		\$289.00	\$240.00
			\$289.00	\$240.00
102-130-017-000	Single Equivalent Dwelling Unit		\$289.00	\$240,00

Assessor's Parcel Number	Property Classification	Number of EDUs	FY 2023-24 Maximum Special Tax	FY 2023-24 Special Tax Levy
	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-130-018-000 102-130-019-000	Single Equivalent Dwelling Unit Single Equivalent Dwelling Unit	an sartī	\$289.00 \$289.00	\$240.00
Not are a substitute of the su	TO LEAVE LANGE CO. LANGE OF THE PROPERTY OF A PROPERTY OF THE		\$289.00	\$240.00
102-130-020-000	Single Equivalent Dwelling Unit Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-130-021-000	Single Equivalent Dwelling Unit Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-130-022-000	Single Equivalent Dwelling Unit Single Equivalent Dwelling Unit	e enganta cenc	\$289.00	\$240.00
102-130-023-000	Single Equivalent Dwelling Unit	. Malail 11 11 11 11 11 11 11 11 11 11 11 11 11	\$289.00	\$240.00
102-130-024-000 102-130-025-000	Single Equivalent Dwelling Unit Single Equivalent Dwelling Unit	aranairean	\$289.00	\$240.00
102-130-025-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-130-020-000	Single Equivalent Dwelling Unit Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-130-027-000	Single Equivalent Dwelling Unit	0	\$289.00	\$240.00
102-130-028-000	Single Equivalent Dwelling Unit	erenings	\$289.00	\$240.00
102-130-029-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-130-030-000	Single Equivalent Dwelling Unit Single Equivalent Dwelling Unit	a di muze h ovens	\$289,00	\$240.00 \$240.00
the Sharman Sharman Sharman and Albanda	Single Equivalent Dwelling Unit	(บุ.ศ. อ.ศ. คระสมเด็จสมเด็จ (คร.) 1	\$289.00	\$240.00
102-130-032-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-130-035-000	Single Equivalent Dwelling Unit Single Equivalent Dwelling Unit	- ::::::::::::::::::::::::::::::::::::	\$289,00	\$240.00
102-130-040-000 102-130-041-000	Single Equivalent Dwelling Unit	uranegira'a	\$289.00	\$240.00
The program of the first of the contract of th	Single Equivalent Dwelling Unit	. 15 61. 1 . 1 . 1	\$289.00	\$240.00
102-130-042-000 102-130-043-000	Single Equivalent Dwelling Unit	i di karinga	\$289,00	\$240.00
102-130-043-000	Single Equivalent Dwelling Unit Single Equivalent Dwelling Unit	n og i skakta t ogskriv 1	\$289.00	\$240.00
grand production of a grand of the contract of the second	Single Equivalent Dwelling Unit Single Equivalent Dwelling Unit	re araîsan	\$289.00	\$240.00
102-130-045-000 102-130-046-000	Single Equivalent Dwelling Unit	on mala labasa k an salaba 1	\$289.00	\$240.00
102-130-047-000	Single Equivalent Dweiling Unit	n na vystanii argani	\$289.00	\$240,00
and a contract of the contract	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-130-048-000	Single Equivalent Dwelling Unit Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-130-049-000	Single Equivalent Dwelling Unit	r, o too slottiffste wike. 1	\$289.00	\$240.00
102-130-052-000	Single Equivalent Dwelling Unit	r trast (silijania)	\$289,00	\$240.00
102-130-053-000	Single Equivalent Dwelling Unit	lining at the data de-	\$289.00	\$240.00
102-130-054-000 102-130-055-000	Single Equivalent Dweiling Unit	ran aktorin	\$289.00	\$240.00
102-130-056-000	Single Equivalent Dwelling Unit	:	\$289.00	\$240.00
102-130-057-000	Other Multiple Equivalent Dwelling Units		\$578.00	\$480.00
102-140-001-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
Committee of the control of the cont	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-140-002-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-140-003-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-140-006-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-140-007-000			\$289.00	\$240.00
102-140-008-000	Single Equivalent Dwelling Unit	2 ⁻¹ 2-3-4-1	\$289.00	\$240.00
102-140-011-000	Single Equivalent Dwelling Unit Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-140-012-000			\$289.00	\$240.00
102-140-013-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-140-014-000	Single Equivalent Dwelling Unit			\$240.00
102-140-015-000	Single Equivalent Dwelling Unit	and the first of the second	\$289.00	\$240.00
102-140-016-000	Single Equivalent Dwelling Unit	1	\$289.00	
102-140-017-000	Single Equivalent Dwelling Unit	Ţ	\$289.00	\$240.00

Assessor's Parcel Number	Property Classification	Number of EDUs	FY 2023-24 Maximum Special Tax	FY 2023-24 Special Tax Levy
102-140-020-000	Single Equivalent Dwelling Unit		\$289,00	\$240.00
102-140-021-000	Single Equivalent Dwelling Unit		\$289,00	\$240.00
102-140-022-000	Single Equivalent Dwelling Unit	1000	\$289.00	\$240.00
102-140-025-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-140-026-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-140-027-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-140-028-000	Single Equivalent Dwelling Unit	y e e i i i i i i i i i i i i i i i i i	\$289.00	\$240.00
102-140-029-000	Single Equivalent Dwelling Unit	.,	\$289.00	\$240,00
102-140-030-000	Single Equivalent Dwelling Unit	1	\$289,00	\$240.00
102-140-031-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-140-032-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-140-038-000	Single Equivalent Dwelling Unit	istar tamin ali mashabili silim ar 3 1	\$289.00	\$240.00
102-140-041-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-140-044-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-140-045-000	Single Equivalent Dwelling Unit	1005	\$289.00	\$240,00
102-140-048-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-140-049-000	Single Equivalent Dwelling Unit	1	\$289,00	\$240.00
102-140-050-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-140-051-000	Single Equivalent Dwelling Unit		\$289,00	\$240.00
102-140-052-000	Single Equivalent Dwelling Unit	. 125 - 22 - 22 - 23 - 24 - 25 - 25 - 25 - 25 - 25 - 25 - 25	\$289.00	\$240.00
102-140-055-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-140-056-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-140-057-000	Single Equivalent Dwelling Unit	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	\$289.00	\$240.00
102-140-058-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-140-059-000	Single Equivalent Dwelling Unit	r tabiro	\$289.00	\$240.00
102-140-060-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-140-061-000	Single Equivalent Dwelling Unit	10 M 10 10 10 M	\$289.00	\$240.00
102-140-062-000	Single Equivalent Dwelling Unit	1.	\$289.00	\$240.00
102-140-063-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-140-064-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
102-140-067-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
102-140-068-000	Single Equivalent Dwelling Unit	, , , , , , , , , , , , , , , , , , ,	\$289.00	\$240.00
102-140-069-000	Other Multiple Equivalent Dwelling Units	2	\$578,00	\$480.00
102-140-070-000	Other Multiple Equivalent Dwelling Units	2	\$578.00	\$480,00
102-140-071-000	Other Multiple Equivalent Dwelling Units	2	\$578.00	\$480.00
102-140-072-000	Other Multiple Equivalent Dwelling Units	2	\$578.00	\$480.00
102-160-003-000	Other Multiple Equivalent Dwelling Units	7.	\$2,023.00	\$1,680.00
103-010-001-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-010-002-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-010-003-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240,00
103-010-004-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-010-005-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-010-005-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-010-007-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-020-001-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00

Assessor's Parcel Number	Property Classification	Number of EDUs	FY 2023-24 Maximum Special Tax	FY 2023-24 Special Tax Levy
103-020-002-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-020-003-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240,00
103-020-004-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-020-005-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
103-020-006-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-020-007-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
103-020-008-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-020-009-000	Single Equivalent Dwelling Unit		\$289,00	\$240.00
103-020-010-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-020-011-000	Exempt	0	\$0.00	\$0.00
103-020-012-000	Single Equivalent Dwelling Unit	Amalanda keradi Amaherah Sala ada 1	\$289.00	\$240.00
103-020-013-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240,00
103-020-014-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-020-017-000	Single Equivalent Dwelling Unit		\$289,00	\$240.00
103-020-020-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-020-021-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
103-020-022-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240,00
103-020-023-000	Single Equivalent Dwelling Unit	T	\$289.00	\$240.00
103-020-024-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-020-025-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-020-026-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240,00
103-020-027-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
103-020-028-000	Other Multiple Equivalent Dwelling Units	2	\$578.00	\$480.00
103-020-029-000	Other Multiple Equivalent Dwelling Units	2 7 2	\$578.00	\$480.00
103-030-001-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-030-002-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
103-030-003-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-030-004-000	Single Equivalent Dwelling Unit	1 1	\$289.00	\$240.00
103-030-005-000	Single Equivalent Dwelling Unit	1	\$289,00	\$240.00
103-030-006-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-030-007-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-030-010-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
103-030-011-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-030-012-000	Single Equivalent Dwelling Unit	1	\$289,00	\$240.00
103-030-013-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-030-014-000	Single Equivalent Dwelling Unit	7 - 1 11 5	\$289.00	\$240.00
103-030-015-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240,00
103-030-016-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-030-017-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-030-018-000	Single Equivalent Dwelling Unit	i	\$289.00	\$240,00
103-030-019-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-030-020-000	Single Equivalent Dwelling Unit		\$289,00	\$240.00
103-030-021-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-030-022-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-030-023-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00

Assessor's Parcel Number	Property Classification	Number of EDUs	FY 2023-24 Maximum Special Tax	FY 2023-24 Special Tax Levy
103-030-024-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
103-030-025-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-030-026-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-030-027-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240,00
103-030-028-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
103-030-029-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-030-030-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
103-030-031-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-030-032-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240,00
103-030-033-000	Single Equivalent Dwelling Unit	1	\$289,00	\$240,00
103-030-034-000	Single Equivalent Dwelling Unit	1 1	\$289.00	\$240,00
103-030-035-000	Other Multiple Equivalent Dwelling Units	2	\$578,00	\$480.00
103-040-001-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
103-040-004-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-040-005-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240,00
103-040-006-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-040-007-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-040-010-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-040-011-000	Single Equivalent Dwelling Unit	1 7 1 6 2	\$289.00	\$240.00
103-040-012-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-040-013-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-040-014-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-040-015-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-040-016-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-040-017-000	Single Equivalent Dwelling Unit		\$289.00	\$240.00
103-040-018-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-040-019-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-040-020-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-040-021-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-040-022-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-040-023-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-040-024-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-040-025-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-040-026-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-040-027-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-040-028-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-040-029-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-040-030-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-040-031-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-040-032-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-040-033-000	Single Equivalent Dwelling Unit	1	\$289,00	\$240.00
103-040-034-000	Single Equivalent Dwelling Unit	I	\$289.00	\$240.00
103-040-035-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-040-036-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-040-037-000	Single Equivalent Dwelling Unit	1	\$289,00	\$240.00

Assessor's Parcel Number	Property Classification	Number of EDUs	FY 2023-24 Maximum Special Tax	FY 2023-24 Special Tax Levy
103-040-038-000	Single Equivalent Dwelling Unit	1	\$289.00	\$240.00
103-040-039-000	Single Equivalent Dwelling Unit	oversirete	\$289.00	\$240,00
103-040-044-000	Single Equivalent Dwelling Unit	:24-44-57	\$289.00	\$240.00
103-040-045-000	Single Equivalent Dwelling Unit	77.78 i y. 263	\$289.00	\$240.00
103-040-046-000	Single Equivalent Dwelling Unit	rwati da Makinda 1	\$289.00	\$240.00
103-040-047-000	Single Equivalent Dwelling Unit	este si nast	\$289.00	\$240.00
103-040-048-000	Single Equivalent Dwelling Unit	. 20.50 / 10.00 (1	\$289.00	\$240.00
103-040-049-000	Single Equivalent Dwelling Unit	mariner.	\$289.00	\$240.00
103-040-050-000	Single Equivalent Dwelling Unit	t dia dati taka ila 1	\$289.00	\$240.00
103-040-051-000	Other Multiple Equivalent Dwelling Units	777 5 77 7 77	\$578.00	\$480.00
103-040-052-000	Other Multiple Equivalent Dwelling Units	. (146) - 14 7 0 1466 - 1	\$578.00	\$480.00
103-050-001-000	Tract 1017, Phase 2	**************************************	\$0.00	\$0.00
103-050-001-000	Tract 1017, Phase 2	0 0	\$0.00	\$0.00
103-050-002-000	Tract 1017, Phase 2	วาการ ก ้าวประช	\$0.00	\$0.00
103-050-004-000	Tract 1017, Phase 2	ిందం కచ్చింది. గ	\$0.00	\$0.00
103-050-005-000	Tract 1017, Phase 2	ŏ	\$0.00	\$0.00
103-050-005-000	Tract 1017, Phase 2		\$0.00	\$0.00
103-050-007-000	Tract 1017, Phase 2	ะระบารทั้ง	\$0.00	\$0.00
103-050-007-000	Tract 1017, Thase 2	0	\$0.00	\$0.00
103-050-009-000	Tract 1017, Phase 2	าย 17 กั ทย 69	\$0.00	\$0.00
103-050-010-000	Tract 1017, Phase 2		\$0.00	\$0.00
103-050-011-000	Tract 1017, Phase 2	ากรกับเร	\$0.00	\$0.00
103-050-012-000	Tract 1017, Phase 2	o itari i hey italia ila. O	\$0,00	\$0.00
103-050-012-000	Tract 1017, Phase 2	in thins	\$0.00	\$0.00
103-050-014-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-050-015-000	Tract 1017, Phase 2	ra začesta	\$0.00	\$0.00
103-050-016-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-050-017-000	Tract 1017, Phase 2	er er ja nger i	\$0.00	\$0.00
103-050-018-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-050-019-000	Tract 1017, Phase 2	or my Karan	\$0.00	\$0.00
103-050-020-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-050-021-000	Tract 1017, Phase 2	To the contract of the contrac	\$0.00	\$0.00
103-050-022-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-050-023-000	Tract 1017, Phase 2	sasanijanan.	\$0.00	\$0.00
103-050-024-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-050-025-000	Tract 1017, Phase 2	· · · · · · · · · · · · · · · · · · ·	\$0.00	\$0.00
103-060-001-000	Tract 1017, Phase 2	0	\$0,00	\$0.00
103-060-002-000	Tract 1017, Phase 2		\$0.00	\$0.00
103-060-003-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-060-004-000	Tract 1017, Phase 2	ň	\$0.00	\$0.00
103-060-005-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-060-006-000	Tract 1017, Phase 2	and the second	\$0.00	\$0.00
103-060-007-000	Tract 1017, Thase 2	n	\$0.00	\$0.00
103-060-008-000	Tract 1017, Phase 2	n	\$0.00	\$0.00
103-060-009-000	Tract 1017, Phase 2	0	\$0.00	\$0.00

Assessor's Parcel Number	Property Classification	Number of EDUs	FY 2023-24 Maximum Special Tax	FY 2023-24 Special Tax Levy
103-060-010-000	Tract 1017, Phase 2	0.75	\$0.00	\$0.00
103-060-011-000	Tract 1017, Phase 2	lan olik 6 o karinta eza erikuluk tuktifuluk. O	\$0.00	\$0.00
103-060-012-000	Tract 1017, Phase 2	0 4	\$0.00	\$0.00
103-060-013-000	Tract 1017, Phase 2	o di Maria de la	\$0.00	\$0.00
103-060-014-000	Tract 1017, Phase 2	0	\$0.00	\$0,00
103-060-015-000	Tract 1017, Phase 2	o (1) . The second of the seco	\$0.00	\$0.00
103-060-016-000	Tract 1017, Phase 2	0.00	\$0.00	\$0.00
103-060-017-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-060-018-000	Tract 1017, Phase 2	0	\$0.00	\$0,00
103-060-019-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-060-020-000	Tract 1017, Phase 2		\$0.00	\$0.00
103-060-021-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-060-022-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-060-023-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-060-024-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-060-025-000	Tract 1017, Phase 2	0	\$0,00	\$0.00
103-060-026-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-060-027-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-060-028-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-060-029-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-060-030-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-060-031-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-060-032-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-060-033-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-060-034-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-060-035-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-060-036-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-060-037-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-060-038-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-060-039-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-070-001-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-070-002-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-070-003-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-070-004-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-070-005-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-070-006-000	Tract 1017, Phase 2		\$0.00	\$0.00
103-070-007-000	Tract 1017, Phase 2	0	\$0.00	\$0,00
103-070-008-000	Tract 1017, Phase 2		\$0.00	\$0.00
103-070-009-000	Tract 1017, Phase 2		\$0.00	\$0,00
103-070-010-000	Tract 1017, Phase 2		\$0.00	\$0.00
103-070-011-000	Tract 1017, Phase 2		\$0.00	\$0.00
103-070-012-000	Tract 1017, Phase 2		\$0.00	\$0.00
103-070-013-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-070-014-000	Tract 1017, Phase 2		\$0.00	\$0.00
103-070-015-000	Tract 1017, Phase 2		\$0.00	\$0.00

Assessor's Parcel Number	Property Classification	Number of EDUs	FY 2023-24 Maximum Special Tax	FY 2023-24 Special Tax Levy
103-070-016-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-070-017-000	Tract 1017, Phase 2		\$0.00	\$0.00
103-070-018-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-070-019-000	Tract 1017, Phase 2	0.00	\$0.00	\$0.00
103-070-020-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-070-021-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-070-022-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-070-023-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-070-024-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-070-025-000	Tract 1017, Phase 2	0.00	\$0,00	\$0.00
103-070-026-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-070-027-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-070-028-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-070-029-000	Tract 1017, Phase 2		\$0.00	\$0.00
103-070-030-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-070-031-000	Tract 1017, Phase 2	. 0	\$0.00	\$0.00
103-070-032-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-080-001-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-080-002-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-080-003-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-080-004-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-080-005-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-080-006-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-080-007-000	Tract 1017, Phase 2	0	\$0,00	\$0.00
103-080-008-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-080-009-000	Tract 1017, Phase 2	0	\$0,00	\$0.00
103-080-010-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-080-011-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-080-012-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-080-013-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-080-014-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-080-015-000	Tract 1017, Phase 2	0	\$0,00	\$0.00
103-080-016-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-080-017-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-080-018-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-080-019-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-080-020-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-080-021-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-080-022-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-080-023-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-080-024-000	Tract 1017, Phase 2	0	\$0,00	\$0.00
103-080-025-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-090-001-000	Tract 1017, Phase 2	0	\$0,00	\$0.00
103-090-002-000	Tract 1017, Phase 2	0	\$0.00	\$0,00
103-090-003-000	Tract 1017, Phase 2	0	\$0.00	\$0.00

Assessor's Parcel Number	Property Classification	Number of EDUs	FY 2023-24 Maximum Special Tax	FY 2023-24 Special Tax Levy
103-090-004-000	Tract 1017, Phase 2	0 7	\$0.00	\$0.00
103-090-005-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-090-006-000	Tract 1017, Phase 2	0.00	\$0.00	\$0.00
103-090-007-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-090-008-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-090-009-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-090-010-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-090-011-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-090-012-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-090-013-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-090-014-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-090-015-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-100-001-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-100-002-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-100-003-000	Tract 1017, Phase 2	0	\$0,00	\$0,00
103-100-004-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-100-005-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-100-006-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-100-007-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-100-008-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-100-009-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-100-010-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-100-011-000	Tract 1017, Phase 2	0	\$0.00	\$0,00
103-100-012-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-100-013-000	Tract 1017, Phase 2	0	\$0.00	\$0,00
103-100-014-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-100-015-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-100-016-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-100-017-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-100-018-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-100-019-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-100-020-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-100-021-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-100-022-000	Tract 1017, Phase 2	0	\$0.00	\$0,00
103-100-023-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-100-024-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-100-025-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-100-026-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-100-027-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-100-028-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-100-029-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-100-030-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-100-031-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-110-001-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-110-002-000	Tract 1017, Phase 2	0	\$0,00	\$0,00

Assessor's Parcel Number	Property Classification	Number of EDUs	FY 2023-24 Maximum Special Tax	FY 2023-24 Special Tax Levy
103-110-003-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-110-004-000	Tract 1017, Phase 2		\$0.00	\$0.00
103-110-005-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-110-006-000	Tract 1017, Phase 2	(\$0.00	\$0.00
103-110-007-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-110-008-000	Tract 1017, Phase 2		\$0,00	\$0.00
103-110-009-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-110-010-000	Tract 1017, Phase 2	0 (0	\$0.00	\$0.00
103-110-011-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-110-012-000	Tract 1017, Phase 2	0	\$0,00	\$0.00
103-110-013-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-110-014-000	Tract 1017, Phase 2	0	\$0,00	\$0.00
103-110-015-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-110-016-000	Tract 1017, Phase 2	Ö	\$0.00	\$0.00
103-110-017-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-110-018-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-110-019-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-110-020-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-110-021-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-110-022-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-110-023-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-110-024-000	Tract 1017, Phase 2	0	00,00	\$0.00
103-110-025-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-110-026-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-120-001-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-120-002-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-120-003-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-120-004-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-120-005-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-120-006-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-120-007-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-120-008-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-120-009-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-120-010-000	Tract 1017, Phase 2	0	\$0,00	\$0.00
103-120-011-000	Tract 1017, Phase 2		\$0.00	\$0.00
103-120-012-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-120-013-000	Tract 1017, Phase 2		\$0.00	\$0.00
103-120-014-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-120-015-000	Tract 1017, Phase 2	., 0	\$0.00	\$0.00
103-120-016-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-120-017-000	Tract 1017, Phase 2		\$0.00	\$0.00
103-120-018-000	Tract 1017, Phase 2		\$0.00	\$0.00
103-120-019-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-120-020-000	Tract 1017, Phase 2		\$0.00	\$0.00
103-120-021-000	Tract 1017, Phase 2	0	\$0.00	\$0.00

Assessor's Parcel Number	Property Classification	Number of EDUs	FY 2023-24 Maximum Special Tax	FY 2023-24 Special Tax Levy
103-120-022-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-120-023-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-120-024-000	Tract 1017, Phase 2		\$0.00	\$0,00
103-120-025-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-120-026-000	Tract 1017, Phase 2		\$0.00	\$0.00
103-120-027-000	Tract 1017, Phase 2	0	\$0,00	\$0.00
103-130-001-000	Tract 1017, Phase 2	0	\$0,00	\$0.00
103-130-002-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-130-003-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-130-004-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-130-005-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-130-006-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-130-007-000	Tract 1017, Phase 2	0	\$0.00	\$0,00
103-130-008-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-130-009-000	Tract 1017, Phase 2	0	\$0.00	\$0,00
103-130-010-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-130-011-000	Tract 1017, Phase 2	0	\$0,00	\$0.00
103-140-001-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-140-002-000	Tract 1017, Phase 2	φ	\$0.00	\$0.00
103-140-003-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-140-004-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-140-005-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-140-006-000	Tract 1017, Phase 2	0	\$0,00	\$0.00
103-140-007-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-140-008-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-140-009-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-140-010-000	Tract 1017, Phase 2	0	\$0.00	\$0,00
103-140-011-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-140-012-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-150-002-000	Exempt	0	\$0.00	\$0.00
103-150-004-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-150-005-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-150-006-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-150-007-000	Tract 1017, Phase 2	0	\$0.00	\$0.00
103-150-009-000	Tract 1017, Phase 2	0	\$0.00	\$0.00

Total Fiscal Year 2023-24 Special Tax Levy \$410,958.00 \$341,280.00

Goodwin Consulting Group, Inc.

APPENDIX C Rate and Method of Apportionment of Special Tax

Ехнівіт В

RIO ALTO WATER DISTRICT COMMUNITY FACILITIES DISTRICT NO. 2011-1 (WASTEWATER SYSTEM IMPROVEMENT PROJECT)

RATE AND METHOD OF APPORTIONMENT OF SPECIAL TAX

Special Taxes applicable to each Assessor's Parcel in the Rio Alto Water District Community Facilities District No. 2011-1 (Wastewater System Improvement Project) shall be calculated, levied, and collected according to the tax liability determined by the Administrator through the application of the appropriate amount or rate for Taxable Property, as described below. All of the property in CFD No. 2011-1, unless exempted by law or by the provisions of Section F below, shall be taxed for the purposes, to the extent, and in the manner herein provided, including property subsequently annexed to CFD No. 2011-1 unless a separate Rate and Method of Apportionment of Special Tax is adopted for the annexation area.

A. <u>DEFINITIONS</u>

The terms hereinafter set forth have the following meanings:

- "Act" means the Mello-Roos Community Facilities Act of 1982, as amended, being Chapter 2.5, (commencing with Section 53311), Division 2 of Title 5 of the California Government Code.
- "Administrative Expenses" means any or all of the following actual or reasonably estimated costs directly related to the administration of CFD No. 2011-1: the costs of computing the Special Taxes and preparing the annual collection schedules (whether by the Water District or any designee thereof or both), the costs of collecting the Special Taxes, the fees and expenses of legal counsel, charges levied by the County in connection with the levy and collection of Special Taxes, costs related to responding to public inquiries, protests, or appeals regarding the Special Taxes, costs associated with complying with any continuing disclosure requirements with respect to the Obligations and the Special Taxes (if applicable), attorney's fees and other costs related to commencing and pursuing to completion any foreclosure of delinquent Special Taxes, and all other costs and expenses of the Water District and County in any way related to the establishment or administration of the CFD.
- "Administrator" shall mean the person or firm designated by the Water District to administer the Special Taxes according to this RMA.
- "Annual Special Tax" means an annual special tax calculated and levied on each Parcel of Taxable Property for any Fiscal Year to pay the Special Tax Requirement, as defined below, subject to the limitation that the Annual Special Tax to be levied on any Parcel of Taxable Property cannot exceed the Maximum Annual Special Tax applicable to that Parcel.

- "Application for Sewer Service" means the application submitted by the owner of a Parcel to connect one or more structures on the Parcel to the Water District's sewer system.
- "Assessor's Parcel" or "Parcel" means a lot or parcel shown on a County Assessor's Parcel map with an assigned County Assessor's Parcel number.
- "Authorized Facilities" means the public facilities authorized to be financed, in whole or in part, by Special Tax proceeds and/or proceeds of sale of Obligations.
- "Board" means the Board of Directors of the Rio Alto Water District.
- "Buildable Lot" means an individual lot within the CFD for which a building permit may be issued without further subdivision of such lot.
- "Catch-Up Special Tax" means a one-time special tax paid by a Parcel of Optional LPSS Property or Optional Septic Property prior to approval by the Water District of the property owner's Application for Sewer Service.
- "CFD" or "CFD No. 2011-1" means the Rio Alto Water District Community Facilities District No. 2011-1 (Wastewater System Improvement Project).
- "CFD Formation" means the date on which the Resolution of Formation relating to the CFD was adopted by the Board.
- "County" means the County of Tehama.
- **"EDU"** means the equivalent dwelling unit factor assigned to each Parcel of Taxable Property for purposes of apportioning the Maximum Annual Special Tax.
- "Fiscal Year" means the period starting July 1 and ending on the following June 30.
- "Halfplex" means a Residential Unit that shares a common wall with another Residential Unit and is assigned its own unique Assessor's Parcel number.
- "Maximum Annual Special Tax" means the greatest amount of Annual Special Tax that can be levied on an Assessor's Parcel in any Fiscal Year determined in accordance with Section C below.
- "Multiple EDU Parcel" means a Parcel of Taxable Property that is assigned two or more EDUs.
- "Obligations" means bonds, loans, or other debt (as defined in the Act), whether in one or more series, issued, insured, or assumed by the CFD related to public infrastructure and/or improvements that are authorized to be funded by the CFD.
- "Optional LPSS Property" means any Parcel within the boundaries of the CFD that is currently connected to, or has the option to connect to, a private septic system, but also has the option to connect to a private low pressure sewer system at a later date, as determined by the Water District.

- "Optional Septic Property" means any Parcel within the boundaries of the CFD that is currently connected to, or has the option to connect to, a private septic system, but also has the option to connect to the Water District's sewer system at a later date, as determined by the Water District.
- "Property Owner Association" or "POA" means a property owner association, including any master or sub-association, existing within the CFD.
- "Property Owner Association Property" or "POA Property" means, for each Fiscal Year, any property within the boundaries of CFD No. 2011-1 that was owned by a Property Owner Association as of January 1 of the prior Fiscal Year.
- "Public Property" means any property within the boundaries of the CFD that is owned by or irrevocably offered for dedication to the federal government, State of California, County, or other local government or public agency.
- "Residential Property" means, in any Fiscal Year, a Parcel for which a building permit was issued on or prior to June 30 of the preceding Fiscal Year for the construction of one or more Residential Units.
- "Residential Unit" means an individual single-family detached unit; an individual residential unit within a duplex, Halfplex, triplex, fourplex, townhome, live/work or condominium structure; an individual mobile home; or an individual apartment unit.
- "RMA" means this Rate and Method of Apportionment of Special Tax.
- "Special Taxes" means, collectively, the Annual Special Tax and the Catch-Up Special Tax.
- "Special Tax Requirement" means the amount necessary in any Fiscal Year (i) to pay principal and interest, including scheduled sinking fund payments, on the Obligations, (ii) pay Administrative Expenses of CFD No. 2011-1, (iii) to create or replenish reserve funds, subject to the 10% limitation set forth in Section D below, (iv) to cure any delinquencies in the payment of principal or interest on indebtedness of CFD No. 2011-1 which have occurred in the prior Fiscal Year or (based on delinquencies in the payment of the Annual Special Taxes which have already taken place) are expected to occur in the Fiscal Year in which the tax will be collected, subject to the 10% limitation set forth in Section D below, and (v) to construct or acquire Authorized Facilities. The amounts referred to in clauses (i) and (ii) of the preceding sentence may be reduced in any Fiscal Year by: (i) interest earnings on or surplus balances in funds and accounts for the Obligations to the extent that such earnings or balances are available to apply against debt service pursuant to legal documents that set forth these terms; (ii) proceeds received by CFD No. 2011-1 from the collection of penalties associated with delinquent Annual Special Taxes; and (iii) any other revenues available to pay debt service on the Obligations as determined by the Administrator.
- "Single EDU Parcel" means a Parcel of Taxable Property that is assigned one EDU.

"Taxable Property" means all of the Assessor's Parcels within the boundaries of the CFD which are not exempt from the Special Taxes pursuant to law or Section F below.

"Taxable Property Owner Association Property" or "Taxable POA Property" means:

- 1. The following four Parcels that were owned by the POA at CFD Formation and not expected to be sold for residential development:
 - a. APN 009-080-36 (Steelhead Landing)
 - b. APN 102-010-11 (Parcel above Steelhead)
 - c. APN 102-100-73 (Park on Edgewater)
 - d. APN 102-120-60 (Park on North/South Marina); and
- 2. APN 102-070-12 (Lake Club); and
- 3. Any Buildable Lot owned by the POA that is expected to be resold to an individual property owner.

"Taxable Public Property" means, in any Fiscal Year, any Parcel of Public Property that had in its entirety been a Parcel of Taxable Property in a prior Fiscal Year.

"Tract 1017, Phase 2 Property" means any Parcel within Phase 2 of Tract Map 1017 at CFD Formation, as identified in the CFD boundary map.

"Water District" means the Rio Alto Water District.

B. ASSIGNMENT OF EDUS

1. Single EDU Parcels

For purposes of levying Special Taxes pursuant to this RMA, the following Parcels shall be categorized as Single EDU Parcels and shall be assigned one EDU:

- a. Except as otherwise provided in any subsection of this Section B, Residential Property on which one single family detached unit has been constructed.
- b. Each Parcel with a Halfplex.
- c. Any Parcel of Taxable Property that is a Buildable Lot on which no structure has yet been constructed.
- d. The following Assessor's Parcels which, at the time of CFD Formation, were Taxable POA Property:
 - (i) APN 009-080-36 (Steelhead Landing)
 - (ii) APN 102-010-11 (Parcel above Steelhead)

- (iii) APN 102-100-73 (Park on Edgewater)
- (iv) APN 102-120-60 (Park on North/South Marina)

2. Multiple EDU Parcels

For purposes of levying Special Taxes pursuant to this RMA, the following Parcels shall be categorized as Multiple EDU Parcels with the following EDU assignments:

- a. A Parcel of Residential Property with a duplex shall be assigned two EDUs.
- b. A Parcel of Residential Property with a triplex shall be assigned three EDUs.
- c. A Parcel of Residential Property with a fourplex shall be assigned four EDUs.
- d. Any Parcel with a residential structure with five or more Residential Units shall be assigned a number of EDUs equal to the number of Residential Units on the Parcel.
- e. The following Assessor's Parcels shall be assigned two EDUs:
 - (i) APN 102-070-12 (Lake Club)
 - (ii) APN 100-300-26 (3 or more merged lots)
 - (iii) APN 102-055-22 (3 or more merged lots)
 - (iv) APN 102-100-71 (3 or more merged lots)
 - (v) APN 102-120-55 (3 or more merged lots)
- f. If two or more Parcels are combined to form one or more new Parcels, the aggregate number of EDUs that had been assigned to the original Parcels shall be assigned to the newly-created Parcel(s).
- g. Any Parcel of Residential Property with more than one Residential Unit may be a Multiple EDU Parcel as determined by the Water District based on services requested for each Residential Unit.
- h. Parcels with commercial or other non-residential buildings shall be assigned two EDUs unless the Water District determines a different number of EDUs is more appropriate based on the type of non-residential building constructed.

If a Parcel is subdivided to create two or more Parcels, the newly-created Parcels shall be assigned EDUs by the Water District based on their planned use. The Water District shall have full discretion in assigning EDUs to newly-created Parcels, and at no time shall a subdivision, reconfiguration or consolidation of Parcels in the CFD result in a reduction in the total number of EDUs that were assigned to the Parcel or Parcels being subdivided, reconfigured, or consolidated.

C. MAXIMUM SPECIAL TAXES

1. Annual Special Tax

The Maximum Annual Special Tax for each Parcel of Taxable Property in CFD No. 2011-1 shall be \$289 per EDU.

2. Catch-Up Special Tax

The Catch-Up Special Tax for any Parcel of Optional LPSS Property or Optional Septic Property that is submitting an Application for Sewer Service shall be calculated as follows:

- Step 1: Determine the actual Annual Special Tax per EDU that was levied in all prior Fiscal Years and sum the annual amounts to calculate the total amount levied to date for a single EDU. If the Annual Special Tax levy for the CFD for the current Fiscal Year has already been submitted to the County for placement on the secured tax roll and an Annual Special Tax is not being levied on such Parcel for the current Fiscal Year, then the Annual Special Tax per EDU for the current Fiscal Year shall also be included in the total amount levied per EDU.
- Step 2: Determine the number of EDUs assigned to the Parcel of Optional LPSS Property or Optional Septic Property for which the Catch-Up Special Tax is being calculated.
- Step 3: Multiply the amount determined in Step 1 by the number of EDUs determined in Step 2 to determine the Catch-Up Special Tax for the Parcel.

D. METHOD OF LEVY OF THE ANNUAL SPECIAL TAX

Not later than the 15th day of July in each Fiscal Year, the Administrator shall perform the following:

- Step 1: Determine the classification of each Parcel within the CFD for purposes of assigning EDUs to each Parcel for that Fiscal Year.
- Step 2: Determine the amount of the Maximum Annual Special Tax for each Parcel of Taxable Property, taking into account (i) the amount of the Maximum Annual Special Tax per EDU, as set forth in Section C of this RMA, and (ii) the number of EDUs assigned to each Parcel, and then determine the aggregate amount of the Maximum Annual Special Tax for all Parcels of Taxable Property for that Fiscal Year.

- Step 3: Determine the Special Tax Requirement for that Fiscal Year, subject to the limitation that the amount of the Special Tax Requirement cannot exceed the aggregate amount of the Maximum Annual Special Tax for all Parcels of Taxable Property as determined in Step 2.
- Determine the amount of the Annual Special Tax to be levied for that Fiscal Year on each Parcel of Taxable Property by distributing the Special Tax Requirement among all of the Parcels of Taxable Property in proportion to the number of EDUs assigned to each such Parcel, subject to the limitations that (i) the amount of the Annual Special Tax on each Parcel cannot exceed the Maximum Annual Special Tax for such Parcel, and (ii) in no case shall the amount allocated to any Parcel of Residential Property be more than ten percent (10%) above what the amount would otherwise have been had nothing been included in the Special Tax Requirement on account of delinquencies or defaults in the payment of the Annual Special Tax by the owner or owners of any other Parcel or Parcels.

E. MANNER OF COLLECTION OF SPECIAL TAXES

The Annual Special Tax shall be collected in the same manner and at the same time as ordinary ad valorem property taxes, provided, however, that prepayments are permitted as set forth in Section G below and provided further that the Water District may directly bill, collect at a different time or in a different manner, and/or collect delinquent Annual Special Taxes through foreclosure or other available methods.

The Catch-Up Special Tax shall be collected from Parcels of Optional LPSS Property and Optional Septic Property prior to approval by the Water District of the property owner's Application for Sewer Service.

The Special Taxes shall be levied and collected until principal and interest on Obligations have been repaid and Authorized Facilities to be constructed directly from proceeds of Special Taxes have been completed. However, in no event shall the Special Taxes be levied after Fiscal Year 2055-2056.

F. EXEMPTIONS

Notwithstanding any other provision of this RMA, no Special Taxes shall be levied on the following:

- (i) Public Property, except Taxable Public Property, as defined herein.
- (ii) POA Property, except Taxable POA Property, as defined herein.
- (iii) Parcels of Optional LPSS Property. Once a Parcel of Optional LPSS Property has connected to the Water District's sewer system, it shall thereafter be classified as Taxable Property. The Water District shall identify such Parcel as either a Single EDU Parcel or a Multiple EDU Parcel and assign EDUs accordingly.
- (iv) Parcels of Optional Septic Property. Once a Parcel of Optional Septic Property has connected to the Water District's sewer system, it shall thereafter be classified as

- Taxable Property. The Water District shall identify such Parcel as either a Single EDU Parcel or a Multiple EDU Parcel and assign EDUs accordingly.
- (v) Parcels of Tract 1017, Phase 2 Property. All Parcels of Tract 1017, Phase 2 Property shall be exempt until such time as the Water District accepts water and sewer improvements for property within Phase 2 of Tract 1017. Once such water and sewer improvements have been accepted by the Water District, those Parcels of Tract 1017, Phase 2 Property which can be served by such improvements shall thereafter be classified as Taxable Property. The Water District shall identify such Parcels as either a Single EDU Parcel or a Multiple EDU Parcel and assign EDUs accordingly.
- (vi) Parcels owned by a public utility for an unmanned facility.
- (vii) Parcels subject to an easement that precludes any other use on the Parcels.
- (viii) Any Parcel that has fully prepaid the Annual Special Tax obligation assigned to such Parcel pursuant to the formula set forth in Section G below.

G. PREPAYMENT OF ANNUAL SPECIAL TAX

The following definition applies to this Section G:

"Outstanding Obligations" means all Obligations that remain outstanding, with the following exception: if an Annual Special Tax has been levied against, or already paid by, an Assessor's Parcel making a prepayment, and a portion of the Annual Special Tax will be used to pay a portion of the next principal payment on the Obligations that remain outstanding (as determined by the Administrator), that next principal payment shall be subtracted from the total principal that remains outstanding, and the difference shall be used as the amount of Outstanding Obligations for purposes of this prepayment formula.

The Annual Special Tax obligation applicable to an Assessor's Parcel in the CFD may be prepaid and the obligation of the Assessor's Parcel to pay the Annual Special Tax permanently satisfied as described herein, provided that a prepayment may be made only if there are no delinquent Special Taxes with respect to such Assessor's Parcel at the time of prepayment. An owner of an Assessor's Parcel intending to prepay the Annual Special Tax obligation shall provide the Water District with written notice of intent to prepay. Within 30 days of receipt of such written notice, the Water District or its designee shall notify such owner of the prepayment amount for such Assessor's Parcel.

1. Prior to Entering into Obligations

If a property owner requests to prepay their Annual Special Tax obligation prior to Obligations being entered into by the CFD, the prepayment amount shall be calculated as follows:

Step 1: Determine the Maximum Annual Special Tax that could be collected from the Assessor's Parcel prepaying the Annual Special Tax in the Fiscal Year in which prepayment would be received by the Water District.

- Step 2: Divide the Maximum Annual Special Tax computed pursuant to Step 1 by the Maximum Annual Special Tax revenues that could be collected in that Fiscal Year from property in the entire CFD.
- Step 3: Multiply the quotient computed pursuant to Step 2 by the expected net proceeds of Obligations that will be secured by Annual Special Taxes to pay for Authorized Facilities, as determined by the Water District. The amount determined shall be the Annual Special Tax prepayment amount.

2. After Entering into Obligations

If a property owner requests to prepay their Annual Special Tax obligation after Obligations have been entered into by the CFD, the Prepayment Amount shall be calculated as follows (capitalized terms as defined below):

Redemption Amount

plus Administrative Fees and Expenses

<u>less</u> <u>Reserve Fund Credit</u> equals <u>Prepayment Amount</u>

As of the proposed date of prepayment, the Prepayment Amount shall be determined by application of the following steps:

- Step 1: Determine the Maximum Annual Special Tax that could be collected from the Assessor's Parcel prepaying the Annual Special Tax in the Fiscal Year in which prepayment would be received by the Water District.
- Step 2: Divide the Maximum Annual Special Tax computed pursuant to Step 1 by the Maximum Annual Special Tax revenues that could be collected in that Fiscal Year from property in the entire CFD.
- Step 3: Multiply the quotient computed pursuant to Step 2 by the Outstanding Obligations to compute the amount of Outstanding Obligations to be retired and prepaid (the "Redemption Amount").
- Step 4: Determine the costs of computing the prepayment amount, redeeming Obligations, and recording any notices to evidence the prepayment and redemption (the "Administrative Fees and Expenses").
- Step 5: If and to the extent so provided, a reserve fund credit shall be calculated as a reduction in the applicable reserve fund for the Outstanding Obligations to be redeemed pursuant to the prepayment (the "Reserve Fund Credit").

- Step 6: The Annual Special Tax prepayment is equal to the sum of the amounts computed pursuant to Steps 3 and 4, less the amount computed pursuant to Step 5 (the "Prepayment Amount").
- Step 7: From the Prepayment Amount, the amount computed pursuant to Step 3 shall be deposited into the appropriate fund as established in the legal documents for the Obligations and be used to retire Outstanding Obligations or make debt service payments. The amount computed pursuant to Step 4 shall be retained in the account or fund that is established to pay administrative expenses of the CFD.

Once a prepayment has been received, a Notice of Cancellation of Special Tax Lien shall be recorded against the Parcel. However, a Notice of Cancellation of Special Tax Lien shall not be recorded until all Special Taxes levied on the Parcel in the current or prior Fiscal Years have been collected.

H. INTERPRETATION OF SPECIAL TAX FORMULA

The Water District may interpret, clarify, and revise this RMA to correct any inconsistency, vagueness, or ambiguity, by resolution and/or ordinance, that does not materially affect the levy and collection of the Special Taxes and any security for any Obligations.

I. APPEAL OF SPECIAL TAX LEVY

Any property owner claiming that the amount or application of the Special Taxes (and/or assignment of EDUs) is incorrect may file a written notice of appeal with the Administrator no later than one calendar year after having paid the Special Taxes that are being disputed. The Administrator shall promptly review the appeal within 30 days of receipt of such written notice, and if necessary, meet with the property owner, consider written and oral evidence regarding the amount or application of the Special Taxes, and decide the appeal. If the property owner disagrees with the Administrator's decision, the property owner may then file a written appeal with the Board, whose decision shall be final and conclusive. If the decision of the Administrator (if the appeal is not filed with the Board) or the Board (if the appeal is filed with the Board) requires the Special Taxes to be modified in favor of the property owner, no cash shall be refunded for prior years' Special Taxes (unless it is determined that such Parcel will not have an Annual Special Tax Levy in future Fiscal Years, in which case a cash refund shall be given from the proceeds of the following Fiscal Year's Annual Special Tax collection). Rather, an adjustment shall be made to the next Annual Special Tax levy(ies) as needed. This procedure shall be exclusive and its exhaustion by any property owner shall be a condition precedent to any legal action by such property owner.

APPENDIX D

Boundary Map of CFD No. 2011-1

